

Evaluation of the High Altitude Lidar Observatory (HALO) methane retrievals during the summer 2019 ACT-America campaign

Rory A. Barton-Grimley¹, Amin R. Nehrir¹, Susan A. Kooi², James E. Collins², David B. Harper¹, Anthony Notari¹, Joseph Lee², Joshua P. DiGangi¹, Yonghoon Choi², and Kenneth J. Davis³

¹NASA Langley Research Center, Hampton, VA, USA

²Science Systems and Applications, Inc., Hampton, VA, USA

³Department of Meteorology and Atmospheric Science, and Earth and Environmental Systems Institute,

The Pennsylvania State University, University Park, PA, USA

Correspondence: Rory A. Barton-Grimley (rory.a.barton-grimley@nasa.gov) and Amin R. Nehrir (amin.r.nehrir@nasa.gov)

Received: 25 March 2022 – Discussion started: 1 April 2022 Revised: 28 June 2022 – Accepted: 4 July 2022 – Published: 15 August 2022

Abstract. The NASA Langley Research Center High Altitude Lidar Observatory (HALO) is a multi-functional and modular lidar developed to address the observational needs of NASA's weather, climate, carbon cycle, and atmospheric composition focus areas. HALO measures atmospheric H2O mixing ratios, CH₄ mole fractions, and aerosol/cloud optical properties using the differential absorption lidar (DIAL) and high-spectral-resolution lidar (HSRL) techniques. In 2019 HALO participated in the NASA Atmospheric Carbon and Transport – America campaign on board the NASA C-130 to complement a suite of greenhouse gas in situ sensors and provide, for the first time, simultaneous measurements of column CH₄ and aerosol/cloud profiles. HALO operated in 18 of 19 science flights where the DIAL and integrated path differential absorption (IPDA) lidar techniques at 1645 nm were used for column and multi-layer measurements of CH₄ mole fractions, and the HSRL and backscatter techniques were used at 532 and 1064 nm, respectively, for retrievals of aerosol backscatter, extinction, depolarization, and mixing layer heights. In this paper we present HALO's measurement theory for the retrievals of column and multi-layer XCH₄, retrieval accuracy, and precision including methods for bias correction and a comprehensive total column XCH₄ validation comparison to in situ observations. Comparisons of HALO XCH₄ to in situ-derived XCH₄, collected during spiral ascents and descents, indicate a mean difference of 2.54 ppb and standard deviation (SD) of the differences of 16.66 ppb when employing 15 s along-track averaging (< 3 km). A high correlation coefficient of R = 0.9058 was observed for the 11 in situ spiral comparisons. Column XCH₄ measured by HALO over regional scales covered by the ACT-America campaign is compared against in situ CH₄ measurements carried out within the planetary boundary layer (PBL) from both the C-130 and B200 aircraft. Favorable correlation between the in situ point measurements within the PBL and the remote column measurements from HALO elucidates the sensitivity of a column-integrating lidar to CH₄ variability within the PBL, where surface fluxes dominate the signal. Novel capabilities for CH₄ profiling in regions of clear air using the DIAL technique are presented and validated for the first time. Additionally, profiling of CH₄ is used to apportion the PBL absorption from the total column and is compared to previously reported IPDA cloud slicing techniques that estimate PBL columns using strong echoes from fair weather cumulus. The analysis presented here points towards HALO's ability to retrieve accurate and precise CH₄ columns with the prospects for future multilayer profiling in support of future suborbital campaigns.

1 Introduction

Atmospheric methane (CH₄) is a prominent greenhouse gas (GHG) with an increasingly important role in climate change due to rising emissions and their subsequent impact on radiative forcing. CH₄ has a global warming potential estimated to be 84 and 28 times greater than carbon dioxide (CO₂) over a 20- and 100-year period, respectively (Myhre

et al., 2013). Since pre-industrial times, CH₄ mole fractions have risen by 150 % (Myhre et al., 2013) with the addition of anthropogenic sources identified as the cause of the rising abundance (Dean et al., 2018). CH₄ emissions can be apportioned between anthropogenic influences, such agriculture, waste management (Nisbet et al., 2016; Schaefer et al., 2016), and fossil fuel activities (Maasakkers et al., 2016; Alvarez et al., 2018), and natural sources which are dominated by wetlands (Bousquet et al., 2006, 2011; Schaefer et al., 2016). Though the major sources of atmospheric CH_4 have been identified, uncertainty in emission rates (Ehhalt et al., 2001; Lu et al., 2022) detrimentally affects our understanding of the total CH₄ burden and its subsequent climate impact (Nisbet et al., 2014). Additionally, Lu et al. (2022) indicate that the time and spatial evolution of different emission sectors varies significantly across North America, showing the need for continued atmospheric observations. The relative contributions and strengths of these highly varied sources require improved observations and increased spatial sampling to quantify these changing emissions.

The National Academies of Sciences, Engineering and Medicine (NASEM) 2017-2027 Decadal Survey for Earth Science and Applications from Space (NASEM, 2018) called for further understanding of the sources and sinks of atmospheric CH₄ and the processes that will affect their future abundances and identified the need for improved measurement capabilities to advance the accuracy of climate models and inform policies that influence anthropogenic emissions. Jacob et al. (2016) discuss prominent methods by which atmospheric CH₄ can be measured from a satellite platform, and the subsequent ability of these models to quantify emissions on regional and global scales is detailed. Passive measurements of column CH₄ from satellites (Frankenberg et al., 2011; Yokota et al., 2009; Hu et al., 2018) have been useful in many applications, such as large coverage inverse analyses (Wecht et al., 2014b; Zhang et al., 2021) and regional emission analyses (Wecht et al., 2014a; Zhang et al., 2020; Varon et al., 2020; Cusworth et al., 2021), the latter of which have been afforded by the high spatial resolutions of the most recently deployed sensors (Veefkind et al., 2012; Jervis et al., 2021). Despite the successes of these passive sensors, they are limited to daytime operation, have broad weighting functions that limit understanding of near-surface fluxes, and suffer contamination from clouds, aerosols, and rapid changes in topography.

In situ measurements have been used extensively for quantifying methane emissions. Useful accuracy and precision have been achieved when measuring emissions from cities (Cui et al., 2015; McKain et al., 2015; Heimburger et al., 2017; Plant et al., 2019; Lopez-Coto et al., 2020), and oil and gas production basins (Alvarez et al., 2018; Barkley et al., 2019a), with an emerging ability to track emissions changes over time (Lyon et al., 2021; Lin et al., 2021). The in situ measurement density available for this quality of emissions quantification, however, is limited at present to a small number of intensive study areas (Richardson et al., 2017; Verhulst et al., 2017; Karion et al., 2020). Globalscale (Schuldt et al., 2021) and continental-scale (Andrews et al., 2014) data collections exist, but their density limits the resolution and accuracy of inverse flux estimates (Bousquet et al., 2006; Bruhwiler et al., 2014). Spatially dense observations from aircraft (Barkley et al., 2019b, 2021; Yu et al., 2021) exist and provide a robust dataset that has great potential for improving quantitation of methane emissions; however their extent is limited to point altitude estimates.

Active sensing of atmospheric CH_4 can overcome many of the challenges that limit passive CH_4 and other GHG retrievals. Light detection and ranging (lidar) measurements of GHGs benefit from the direct generation of laser light to enable monitoring in all seasons, latitudes, during day and night, and allows for accurate measurements in the presence of clouds, aerosols, and topographic variability. Currently, no space instruments employing active techniques for GHG monitoring exist; however, development of the MER-LIN (MEthane Remote sensing Lidar missioN) satellite, anticipated 2027 launch (Ehret et al., 2017), will provide global measurements of CH_4 column-averaged dry-air mole fractions (XCH₄) at 1.645 µm.

The differential absorption lidar (DIAL) method (Schotland et al., 1964, 1974) is employed for the measurement of atmospheric CH₄ and other GHGs. At least two wavelengths of laser light are transmitted around a gas absorption line, and differential attenuation through the atmosphere is experienced between the absorbing and non-absorbing wavelengths. The differential attenuation across a prescribed range bin can then be used to directly measure the GHG concentration, where the precision of the measurement is directly proportional to the size of the range bin. The integrated path differential absorption (IPDA) technique, a variation of DIAL, provides high-precision column-averaged dryair mole fractions of a GHG by utilizing strong echoes from clouds and the ground to measure the differential attenuation from the absorbing molecule of interest (Menzies et al., 2003; Ehret et al., 2008). IPDA offers high precision at the expense of profiling and has been demonstrated from airborne platforms as a highly precise and accurate method by which to measure total and partial column abundances of CO₂, CH₄, and other GHGs (Riris et al., 2012, 2017; Dobler et al., 2013; Lin et al., 2015; Abshire et al., 2018; Refaat et al., 2020; Campbell et al., 2020). In preparation for the MERLIN mission, an airborne CH₄ IPDA demonstrator, CHARM-F (Amediek et al., 2017), has made progress towards demonstrating the expected measurement capabilities, targeted error budgets, spectroscopic requirements, and other research necessary to translate an IPDA lidar to spaceborne operation for global CH₄ measurements.

Recently, the NASA Langley Research Center (LaRC) developed a modular airborne DIAL–IPDA lidar to provide multi-functional measurements of GHGs. The High Altitude Lidar Observatory (HALO) was developed as a more capable replacement for the NASA Lidar Atmospheric Sensing Experiment (LASE) H₂O DIAL instrument (Browell et al., 1998) with improved operational flexibility and capability (Nehrir et al., 2018). HALO measures atmospheric H₂O mixing ratios, CH₄ mole fractions, and aerosol/cloud optical properties using the DIAL, IPDA, and high-spectralresolution lidar (HSRL) (Hair et al., 2008) techniques, respectively. HALO was designed as an airborne simulator for future space-borne DIAL-IPDA missions called for by the NASEM Decadal Survey (NASEM, 2018) while also serving as a test bed for risk reduction of key technologies required to enable those future missions. To respond to a wide range of airborne science applications, HALO can be rapidly reconfigured to provide H2O DIAL and HSRL, CH4 DIAL-IPDA and HSRL, or CH₄ DIAL–IPDA and H₂O DIAL measurements using three distinct modular laser transmitters and a single multi-channel and multi-wavelength receiver. First results from the H₂O DIAL and HSRL configuration were discussed in Bedka et al. (2021) and Carroll et al. (2022). Here, we present results from HALO's CH₄ DIAL-IPDA and HSRL configuration, which, to our knowledge, is the first ever demonstration of IPDA-derived XCH₄ with simultaneous HSRL observations of aerosol optical properties. The coincident retrievals of XCH₄ and surrounding environmental contextual information (planetary boundary layer height, PBLH, and aerosol intensive/extensive properties) provide a comprehensive data-generating capability which can be used for constraint of priors for inverse modeling of CH₄ fluxes to enable identification of sources and sinks and inform largescale transport models.

Novel to HALO is the ability to generate profiles of CH₄ differential absorption optical depth (DAOD), in addition to total column DAOD, using the DIAL technique. This retrieval was first demonstrated during the Long Island Sound Tropospheric Ozone Study (Judd et al., 2020). Traditionally this retrieval has been inaccessible to CH₄ IPDA instruments due to weak molecular backscatter at 1645 nm $(\sim 1\%$ of that at 532 nm) and a reduced ability to detect the weakly backscattered light due to poor detector performance at these spectral regions compared to readily available high-gain components available at visible and nearinfrared (NIR) wavelengths. With sufficient along-track averaging, HALO can generate relatively high signal-to-noise ratio (SNR) profiles of backscatter at 1645 nm, allowing access to preliminary range-resolved retrievals. These retrievals have been evaluated for their feasibility and utility in apportioning the PBL region from the total column DAOD in addition to providing an alternate method to retrieve PBL mole fractions in clear-air regions where the cloud slicing technique (Ramanathan et al., 2015; Amediek et al., 2017) cannot be employed. Additionally, profiles of atmospheric backscatter at 1645 nm have been investigated as an alternative method for total column IPDA bias correction without the need for in situ spiral comparisons. The results present here are a preliminary assessment of retrieval performance and their application. A total quantitative assessment of the DIAL technique for CH_4 profiling will require high SNR not currently accessible to HALO due to detector limitations. Improved detector technology, such as advanced HgCdTe detectors (Sun et al., 2017), would enable routine profiling of lower-tropospheric CH_4 for further evaluation and development of higher-level products. Despite lower detector performance, retrievals of IPDA offline atmospheric backscatter have revealed detailed atmospheric structure that could be used for assessment of mixed-layer height (MLH) in lieu of HSRL channels (currently retrieved from the 532 nm HSRL aerosol backscatter). Additionally, the backscatter could be calibrated (Fernald, 1984) to develop new intensive products, such as aerosol wavelength dependence between the 1645 and 1064 nm.

This paper details the first results of HALO's CH₄ DIAL– IPDA and HSRL configuration from the 2019 NASA Atmospheric Carbon and Transport - America (ACT-America) airborne campaign (Davis et al., 2021). It provides a brief overview of the measurement theory, instrument performance, and examples of collocated XCH₄ and HSRL measurements and introduces advanced methods to apportion CH₄ abundances within the planetary boundary layer (PBL) from the column with the DIAL technique. The paper is organized as follows: Sect. 2 provides a brief introduction to the HALO instrument and its measurement approaches. Section 3 gives an overview of the IPDA calibration process, methods to compare column retrievals to in situ validation measurements, bias correction, and performance analysis of XCH₄ precision and accuracy. Section 4 provides examples of retrievals at regional scales with comparison to PBL in situ measurements. Section 5 introduces advanced methods for range-resolved profiling of CH₄ and direct PBL apportionment in clear-air regions. Section 6 summarizes results and provides an outlook towards future impacts of HALO observations.

2 Instrument and retrieval description

2.1 Instrument overview

HALO is a direction detection lidar which employs the DIAL–IPDA, HSRL, and standard backscatter techniques for measurements of GHGs, clouds, and aerosols. The geometry for the combined DIAL–IPDA and HSRL measurement is shown in Fig. 1. HALO is configured such that a single laser transmitter generates all of the requisite wavelengths for the CH₄ DIAL–IPDA (1645 nm), HSRL (532 nm), and backscatter (1064 and 1645 nm) measurements. The laser output is transmitted coaxially with a single collection telescope, from which the backscattered signals are collected and processed with a multi-wavelength receiver that houses conditioning optics, detectors, and control electronics. Specific details of the HALO instrument architecture will be presented in a fu-



Figure 1. HALO measurement geometry from the NASA C-130. Simultaneous acquisition of CH₄ DIAL–IPDA and HSRL data provides information about column CH₄ and aerosols/PBLH, respectively.

ture publication, and the necessary details for retrieval are shown in Table 1.

HALO's CH₄ retrieval is carried out by interrogating the R6 line complex at 1645 nm. The 1 kHz pulse repetition frequency (PRF) laser light at 1645 nm is generated by a tunable optical parametric oscillator (OPO) (Nehrir et al., 2018; Fitzpatrick et al., 2019), which is pumped by a single-frequency injection-seeded Nd:YAG source at 1064 nm (Nehrir et al., 2018). Single-frequency operation of the OPO is achieved by injection-seeding two discrete continuous-wave distributed feedback (DFB) lasers that are spectrally stabilized to the online and offline spectral locations of the R6 line complex, 1645.5518 and 1645.3724 nm, respectively. Injection seeding into the OPO cavity is done using fast electro-optical switches on a shot-to-shot basis, which results in a 500 Hz double pulse repetition frequency output from the OPO. The residual pump light, left over from the OPO conversion process, is frequency doubled to 532 nm after which the combined 1064 and 532 nm outputs are transmitted coaxially with the OPO output and used for the backscatter and HSRL retrievals. Injection seeding, combined with seed laser stabilization and pulsed laser cavity stabilization, ensures a high spectral purity of > 99.9 % of the OPO and 1064 nm pump and allows high measurement accuracy and low bias. Monitoring the pulsed 1064 and 1645 nm outputs and the peak frequency and width of each pulse in real time during flight operations ensures optimal laser performance.

Figure 2 shows CH₄ absorption cross sections at the R6 line complex calculated from the HITRAN 2016 database (Gordon et al., 2017) at two different pressure altitudes along with the transmitted DIAL–IPDA wavelengths (a Voigt line shape is assumed for all of the analysis presented herein). The online wavelength was selected in the trough of the line



Figure 2. Methane absorption cross sections calculated using a Voigt line shape for a standard atmosphere at 0 and 5 km altitude. The online, 1645.5518 nm, and offline, 1645.3724 nm, wavelengths are shown in red.

complex to provide uniform sensitivity to the lower free troposphere and reduce laser stability requirements, compared to operation at the peak of a single absorption line (Kiemle et al., 2011). The offline wavelength was determined by balancing the optimization of the CH_4 DAOD and minimization of the H_2O DAOD.

Unlike the DIAL technique, which does not require knowledge of the online and offline transmitted pulse energies, the IPDA technique requires accurate knowledge of these relative energy differences to normalize the backscattered signal from a scattering surface and calculate the CH₄ DAOD, which is then used to retrieve XCH₄. To capture the relative energy differences between transmitted pulses, a laser energy monitor (LEM) subsystem samples a fraction of the transmitted beam, breaks speckle between laser shots (discussed further in Sect. 3.1.2), and detects the light with a fiber-coupled InGaAs avalanche photodiode (APD), equivalent to those in the receiver.

The received light is collected by a 0.4 m diameter allmetal telescope, passed through a 0.65 nm interference filter to suppress unwanted solar background, and directed towards specific detection chains using dichroic splitters. The HALO CH₄ receiver chain employs three optical detection channels, one for boresight and two for science. The boresight channel directs a small amount of light to a quadrant PIN photodiode to maintain alignment between the transmit and receive paths, and the remaining light is directed to the science channels. The linear dynamic range of the science channels is increased by splitting the light directed to the science channels such that one channel sees approximately 90 % (high optical) and the second sees 10% (low optical) with separate detectors. The dynamic range is further increased by use of a dual buffered output from each detection chain with variable gain settings that cover a signal range exceeding 20 effective bits at the digitizer, or 60 dB. The largesignal dynamic range allows for measurements over varying albedos, through tenuous clouds, and at varying standoff dis-

Table 1. HALO	parameters	during	ACT-A	merica	2019
---------------	------------	--------	-------	--------	------

Parameter	
Laser type	Fibertek: Nd:YAG-pumped injection-seeded OPO
Laser wavelengths	532 nm, 1064 nm, 1645 nm
Transmitted laser energy	1.0 mJ, 2.5 mJ, 2.5 mJ
Laser PRF (532, 1064, 1645 nm)	1 kHz, 1 kHz, 500 Hz double pulse
Laser pulse width	5 ns (532 nm), 20 ns (1064 nm), 15 ns (1645 nm) FWHM
Spectral purity	532 nm: > 99.98, 1645 nm: > 99.96
Laser beam divergence $(1/e^2)$	0.8 mrad (532 nm), 0.8 mrad (1064 nm), 0.4 mrad (1645 nm)
DIAL-IPDA wavelengths	1645.5518 nm, 1645.3724 nm
HSRL-backscatter wavelengths	532.2929 nm, 1064.5859 nm
DIAL-IPDA vertical sampling rate	120 MHz (1.25 m)
Effective vertical resolution	15 m
Reporting interval	2 Hz (500 shot average 532/1064 nm 250 shot on/off average 1645 nm)
Collection aperture	0.4 m
Field of view	1 mrad (532/1064 nm), 0.5 mrad (1645 nm)

tances from the scattering target without instrument reconfiguration or recalibration. The highest-sensitivity channel, high optical and high electrical (HOHE), is used exclusively for atmospheric profiling at the CH₄ wavelengths, a unique feature of HALO. The remaining channels are utilized for the IPDA retrievals from cloud and surface returns: high optical and low electrical (HOLE) for high-altitude operation and/or low-albedo targets, low optical and high electrical (LOHE) for mid-altitude operation, and low optical and low electrical (LOLE) for low-altitude operation and/or high-albedo targets.

The native vertical resolutions for the DIAL-IPDA and HSRL channels are limited by the transmitted laser pulse widths (Table 1). The backscattered 1645 nm signals are digitized at a 120 MHz sample rate (1.25 m resolution in air) with a detection chain bandwidth of 3 MHz. To ensure that the transient response from the surface and clouds is accurately captured, the 1645 nm signals are retained at the 1.25 m vertical resolution for all IPDA calculations, serving to oversample the return pulse. The backscattered 1064 and 532 nm signals are digitized at the 120 MHz sample rate with a 3 and 40 MHz detection chain bandwidth, respectively. To increase SNR and reduce the output file size, the 532 nm signals are digitally filtered and both the 532 and 1064 nm data are decimated to 15 m vertical resolution. Figure 3 shows the ground return response at 1645 nm for a single 0.5 s profile at 1.25 m vertical resolution, where the HOHE profiling channel is fully saturated while the HOLE, LOHE, and LOLE channels remain on scale for IPDA retrievals.

Because of the high PRF of HALO's pulsed laser, real-time onboard averaging is employed using field-programmable gate arrays (FPGAs) to further reduce the size of the recorded data file. The digitized signals are summed on the FPGA to a 2 Hz rate, resulting in 500 accumulated shots for the 532/1064 nm channels and 250 shots at each wavelength for the 1645 nm channels. Although the data col-



Figure 3. Example of ground return impulse responses for the four IPDA receiver channels for a single 0.5 s profile taken from 8 km. A single channel maintains high sensitivity for atmospheric backscatter. A combination of optical and electrical splits allows for optimization of the dynamic range to allow for sampling of the surface return backscatter over a wide range of aircraft altitudes and surface albedos.

lection interval is 2 Hz, the high-PRF transmitter ensures high pulse overlap exceeding $\sim 94\%$ overlap at high altitudes (10 km) and $\sim 87\%$ overlap at mid-altitudes (5 km), considering a $\sim 200 \text{ m s}^{-1}$ aircraft speed. High pulse overlap minimizes the effect of albedo variations between the online and offline IPDA samples, and additional reduction of albedo variation noise to negligible levels is achieved by employing along-track shot averaging (Amediek et al., 2009).

2.2 XCH₄ IPDA measurement technique

The range-resolved 1645 nm backscattered laser light from the ground and clouds can be interpreted through the lidar equation for hard targets (Grant, 1982). The received power at the digitizer from a target at a surface scattering elevation (SSE) is given by

$$P_{\rm rx}(\lambda, R_{\rm SSE}) = \frac{E_{\rm L}(\lambda)}{t_{\rm eff}} \cdot \left(\eta(\lambda)\beta(\lambda)\frac{A}{R_{\rm SSE}^2}\right)$$
$$\cdot e^{-2\left(\tau_{\rm CH_4}(\lambda, R_{\rm SSE}) + \sum \tau_{\rm g}(\lambda, R_{\rm SSE}) + \tau_{\rm m}(\lambda, R_{\rm SSE}) + \tau_{\rm a}(\lambda, R_{\rm SSE})\right)}$$
$$+ P_{\rm b}(\lambda), \qquad (1)$$

where the transmitted energy per laser pulse is E_{L} (J) and the effective time domain response of the return signal is $t_{\rm eff}$ (s). $\eta(\lambda)$ is a unitless wavelength-dependent system constant that contains instrument efficiencies and all scalar values. $\beta(\lambda, R)$ is the target's reflection coefficient (sr⁻¹) and is equated as $\beta(\lambda) = \rho(\lambda) f(\lambda)$, where $\rho(\lambda)$ is the scatterer's reflectivity and $f(\lambda)$ is the bidirectional reflectance distribution function (sr^{-1}) . The area of the telescope aperture is given by A (m²), R_{SSE} is the range to the scattering surface (m), and $A/R_{\rm SSE}^2$ sets the solid angle of the receiver (assumes full geometric overlap of the transmitter and receiver). The exponential describes the two-way transmittance of laser light through the atmosphere and contains the optical depth terms τ_{CH_4} , τ_g , τ_m , and τ_a , which describe the extinction (absorption and scattering) due to CH₄ absorption, other absorbing gases, non-absorbing molecules, and aerosols. These terms can be understood through the Beer-Lambert law, where the optical depth due to CH₄ and the additional interfering gases over the measurement path is given by $\tau(\lambda, R_{\text{SSE}}) = \int_0^{R_{\text{SSE}}} \sigma(\lambda, r') n(r') dr'$, for a given absorption cross section, σ (cm²), and gas number density, *n* (cm^{-3}) . The background molecular atmosphere and aerosol optical depth are defined by their respective extinction coefficients, $\alpha_m(\lambda)$ and $\alpha_a(\lambda)$ (m⁻¹). The solar background is given by $P_{\rm b}(\lambda)$.

The digitized representation of the received power is proportional to the effective temporal response of the instrument and target. Under the assumption of Gaussian subcomponents, $t_{\rm eff} = \sqrt{(t_{\rm L})^2 + (t_{\rm det})^2 + (t_{\rm tgt})^2}$ and is a geometrical sum of the FWHM temporal responses of the transmitted laser pulse width, $t_{\rm L}$; the detection chain, $t_{\rm det}$; and the scattering target, $t_{\rm tgt}$. The detection chain response is composed of a total system bandwidth ($B_{\rm sys}$), with contributions from the detector and post-detection amplifier, and can be approximated by $t_{\rm det} \approx 1/(3B_{\rm sys})$. The temporal response of the target, $t_{\rm tgt}$, is proportional to the terrain roughness and surface structure. From Eq. (1) the target's total power is estimated by integrating over $t_{\rm eff}$ such that for each wavelength $P_{\rm rx}^{\rm sfc}(\lambda) = \int P_{\rm rx}(\lambda, R') dR'$ is computed and used for IPDA retrievals.

To obtain the desired CH_4 measurement, separate expressions of Eq. (1) can be defined at the online and offline wavelengths and used to solve for the DAOD due to CH_4 as

$$\delta \tau_{\text{CH}_4} = \tau_{\text{CH}_4} (\lambda_{\text{on}}) - \tau_{\text{CH}_4} (\lambda_{\text{off}})$$
$$= \frac{1}{2} \ln \left(\frac{P_{\text{rx}}^{\text{sfc}} (\lambda_{\text{off}})}{P_{\text{rx}}^{\text{sfc}} (\lambda_{\text{on}})} \cdot \frac{E_{\text{L}} (\lambda_{\text{off}})}{E_{\text{L}} (\lambda_{\text{off}})} \right).$$
(2)

Equation (2) assumes that many of the variables from Eq. (1) are equivalent between the DIAL–IPDA wavelengths and cancel such that the DAOD is simply defined by the transmitted and received powers. A derivation of Eq. (2) with no assumptions on the wavelength equivalence of terms can be found in Ehret et al. (2008).

The DAOD can be combined with atmospheric state parameters and a pressure-weighting function to retrieve the column-weighted CH_4 dry-air mole fraction as (Dufour and Bréon 2003; Ehret et al., 2017)

$$XCH_4 = \frac{\delta \tau_{CH_4} - \left(\delta \tau_{H_2O} + \delta \tau_{CO_2}\right)}{\int_{p_a}^{p_{SSE}} w\left(p'\right) dp'},$$
(3)

where $\delta \tau_{CH_4}$ has corrections applied to account for the differential absorption of H₂O and CO₂ (the two main interfering molecules) between the online and offline wavelengths, $\delta \tau_{H_2O}$ and $\delta \tau_{CO_2}$. To calculate $\delta \tau_{H_2O}$ and $\delta \tau_{CO_2}$, the relative humidity from reanalysis and a constant mole fraction of 400 ppm are used. In general, the contribution of DAOD due to CO₂ and H₂O is negligible (on the order of 0.0001 DAOD each), but still accounted for. The reduced impact from $\delta \tau_{H_2O}$ results from optimal offline wavelength selection such that $\delta \tau_{H_2O}$ is minimized near the surface (Refaat et al., 2013).

Equation (3)'s weighting function is a description of the instrument's sensitivity to CH_4 absorption as a function of altitude and is explicitly dependent on the online and offline wavelength selection. At each pressure altitude the weighting function is defined as (Kiemle et al., 2011)

$$w(p) = \frac{\Delta \sigma_{\rm CH_4}}{g\left(m_{\rm dry} + m_{\rm H_2O}q_{\rm H_2O}\right)},\tag{4}$$

where $\Delta\sigma_{CH_4}$ is the CH₄ differential absorption cross section (DCS), *g* is the acceleration due to gravity, m_{dry} is the average mass of a dry-air molecule, m_{H_2O} is the mass of a water molecule, and q_{H_2O} is the water vapor mixing ratio. Integration of Eq. (4) from the aircraft's altitude, p_a , to the SSE, p_{SSE} , gives the weighted average along the observed column. Figure 4 shows an example of a weighting function for HALO's spectroscopy, where near-uniform sensitivity can be seen across the lower troposphere and through the PBL.

HALO's retrievals of XCH₄ are performed along the backscatter profile's slant path. The latitude and longitude of the ground spot for each measurement are realized by performing a geometric transformation from the transmitter to the SSE using the aircraft's global positioning system (GPS) and inertial measurement unit (IMU) data. This provides the



Figure 4. Pressure-weighting function used in the XCH₄ retrieval for the HALO operating wavelengths.

surface pressure estimation at the SSE from the atmospheric state parameters. With an effective vertical range resolution of 15 m sampled at 1.25 m, the alignment of the calculated SSE with the GLOBE digital elevation model (DEM) (Hastings and Dunbar, 1998) shows good agreement at 2 Hz, and geolocation was deemed acceptable (an RMSE of 1.19 m over ocean is seen by HALO). Though HALO oversamples the return pulse, Amediek et al. (2013) showed that it was possible to achieve < 10 m ranging from a 150 m pulse, and Ehret et al. (2008) showed that B_{sys} at the low value of 3 MHz would be sufficient to meet requirements for determination of the ground response.

The basic processing steps required to retrieve XCH₄ are described by the flow diagram in Fig. 5. All calculations are performed from the basic quantities acquired during flight: transmitted power, received backscatter profiles, and the aircraft's IMU attitude and GPS timing information; the latter is used for geolocation of the SSE. The retrieval altitude grid is referenced to mean sea level (MSL) such that 0 m altitude is equivalent to the mean elevation of the sea surface (altitude is used to indicate altitude above MSL). The time series of meteorological data inputs used to retrieve XCH₄ from CH₄ DAOD come from reanalysis. Vertically resolved pressure, temperature, and relative humidity curtains are generated along the GPS defined using NASA's Global Modeling and Assimilation Office's (GMAO) Modern-Era Retrospective analysis for Research and Applications, Version-2 (MERRA-2) (Gelaro et al., 2017). The analysis utilizes the 3 h reanalysis product with all parameters converted to geometric height and vertically interpolated to HALO's resolution. Comparisons of retrievals using MERRA-2 atmospheric state to those using in situ profiles from spiral maneuvers indicate that differences are < 1 ppb. Use of MERRA-2 under



Figure 5. Processing flow for the HALO IPDA XCH₄ retrieval.

normal flight operations includes any atmospheric state error within the XCH₄ retrieval, as expected for retrievals made in all regions without access to in situ profiles. To calculate the CO₂, H₂O, and CH₄ DCS the HITRAN2016 spectroscopic database is used (Gordon et al., 2017) with MERRA-2 pressure and temperature inputs. The DCSs are then used for calculation of the weighting function, DAOD correction terms, and within in situ-derived XCH₄ comparisons. Recent analyses for the MERLIN mission have shown that updates to the spectroscopy used in the XCH₄ retrieval process (Delahaye et al., 2016a, b; Vasilchenko et al., 2019) are required to overcome known biases in the line parameters. This translates to retrieval bias and will be investigated for HALO retrievals in future analysis. The broad effects of spectroscopy errors and the impact on retrievals are discussed in later sections.

2.3 HSRL measurement technique

To provide additional information content and further context to the XCH₄ retrieval, HALO employs the HSRL technique at 532 nm and traditional backscatter at 1064 nm. The methods and implemented architecture leverage developments from prior NASA LaRC HSRL instruments (Hair et al., 2008). HALO utilizes an iodine vapor filter in the instrument's receiver to separate backscatter contributions from the broadened molecular scatter, a few gigahertz in width, and the narrow Mie scatter resulting from aerosols, which maintains nearly the same spectral distribution as the incident laser light, < 100 MHz in width. Utilizing the HSRL technique, aerosol extensive parameters – backscatter and extinction and intensive parameters – aerosol lidar ratio, aerosol depolarization ratio, spectral depolarization ratio, Ångström backscatter coefficient, and aerosol typing can be computed. 4630

Aerosol-derived mixed layer heights are computed from the HSRL vertically resolved aerosol backscatter product according to the methods discussed in Scarino et al. (2014). Explicit description of the HSRL techniques is provided in Hair et al. (2008), and their use in HALO's H₂O configuration is further elaborated in Carroll et al. (2022), which mirrors employment in CH₄ configuration.

3 Airborne measurements during ACT-America 2019

HALO was integrated on the NASA C-130 aircraft in the summer of 2019 for the final ACT-America campaign (Davis et al., 2021; Wei et al., 2021), where sorties were conducted out of Shreveport, LA; Lincoln, NB; and the NASA Wallops Flight Facility, VA. During the campaign HALO's operation was limited to flight altitudes above the PBL to minimize instrument exposure to the harsh temperature and vibration environments associated with increased temperature and turbulence within the PBL. Comparison to in situ instruments at regular intervals throughout the campaign provided a robust evaluation of the accuracy and precision of HALO's CH₄ products.

3.1 Performance analysis

For ACT-America, HALO's DIAL–IPDA modality was operated in two configurations. The first utilized an attenuator in the transmit optical path to dynamically maintain signal linearity in the LOHE channel during flight. This has the effect of maintaining backscatter strength from all expected measurement altitudes and thus minimizing the probability of low-SNR retrievals on the LOHE channel. The second configuration transmitted the full laser power at all measurement altitudes and surface conditions; this configuration was exploratory and intended to exercise the full dynamic range of the receiver while providing a dataset by which to evaluate gain splicing of the different detection channels to account for changes in surface albedo and aircraft altitude.

As the IPDA technique relies on independent measurements of the transmitted pulse energy, accurate knowledge of the differential transmission between the transmit and receive path is required. Near-field scattering effects on the differential transmission are ameliorated to the extent possible by placing the 1645 nm channel's field stop prior to the interference filter, which accounts for the largest source of differential transmission through the receiver (Nehrir et al., 2009). Measurement of the system's differential transmission is made by placing a scattering target in front of the transmit beam and collection aperture such that the receiver path is evenly illuminated without attenuation due to CH₄ absorption. We refer to this method as "zero-path" calibration. Many of these effects, and others not discussed here, were correctable with zero-path calibration, were repeatable over the duration of the mission, and have stayed stable since the initial instrument development. Additionally, we found that the zero-path calibration term was independent of transmit power, allowing a single correction term to be applied throughout the entire campaign for each gain. The zero-path calibrations were calculated for each receiver gain in preand post-campaign testing and removed from flight data to give the CH₄ DAOD as $\delta \tau_{CH_4}^{cal} = \delta \tau_{CH_4} - \delta \tau_{ZP}$. The average zero-path calibration values were 0.2971 (LOHE), 0.3128 (HOLE), and 0.2931 (LOLE). The "cal" superscript will be dropped for simplicity.

3.1.1 IPDA optical depth bias correction

Range-dependent biases between the HALO DAOD and in situ measurement-derived DAOD were observed during pre-campaign test flights. Similar biases of comparable magnitude and trend have also been observed in other airborne pulsed and continuous wave IPDA architectures (Campbell et al., 2020; Amediek et al., 2017; Fix et al., 2020). Studies examining the R6 line complex have shown that spectroscopic uncertainty can manifest itself as systematic bias in the retrieval of CH₄ from a remote sensor (Delayhe et al., 2016a, b, 2019; Vasilchenko et al., 2019). Additional sources of error, such as laser spectral impurity, imprecise knowledge of transmitted wavelength, and other sources of systematic effect (Ismail and Browell, 1989) could potentially contribute to the observed range-dependent bias; however, real-time characterization of the laser performance indicated that the laser transmitter was performing nominally. Sources of bias could also arise from intrinsic errors in the method of lidar to in situ column comparisons, such as temporal phasing of the in situ spiral relative to the lidar overpass (important when comparisons are in or near source regions) and misrepresentation of the total column by the in situ measurements due to the lack of observations at the surface. The latter spiral sampling issues were constraints of the mission, and spectroscopic uncertainty is beyond the scope of this paper. Recent studies have indicated that statistical and geophysical biases can also manifest from low-SNR retrievals or from sufficiently long along-track averaging, though corrections have been developed for each (Tellier et al., 2018). Initial assessment of HALO's native 0.5 s retrievals for each gain channel found that optimized receiver gain exhibiting high SNR displayed negligible CH₄ DAOD statistical bias, $< 1 \times 10^{-3}$ in DAOD.

Test flights at the beginning and end of the campaign were utilized to compare HALO XCH₄ retrievals with in situderived XCH₄ and develop subsequent correction methods to remove the observed systematic bias. Stair-step descent maneuvers were employed followed by a descending spiral between each altitude leg for in situ CH₄ profiles. Each stair step overflew the same ground track to generate multiple HALO DAOD estimates from fixed altitudes while observing the same air mass. A co-located Picarro spectrometer, calibrated to the WMO X2004A scale (DiGangi et al., 2021), on board the C-130 was utilized for in situ observations of the CH₄ mole fraction. The lowest altitude of the spiral, ~ 300 m above ground level (a.g.l.), was filled in by extrapolating the last measurement to the ground to provide a complete profile from max flight altitude to the SSE. The in situ CH₄ mole fraction profile is converted to number density, combined with the HALO DCS (in situ pressure and temperature profiles are utilized) and integrated from the respective altitude of each leg to the SSE. This generates a multi-point set of in situ-derived DAOD estimates from which the analogous HALO measurements can be directly compared to and any bias quantified. The potential impact of near-surface variations in CH₄ was minimized by selecting locations that were distant from known point sources and by restricting maneuvers to the convective BL, such that vertical gradients close to the ground would be minimized.

To generate the bias correction terms, a fractional difference between the mean in situ-derived DAOD, $\delta \overline{\tau}_{IS}$, and the mean HALO DAOD, $\delta \overline{\tau}_{CH_4}$, for each altitude leg is calculated as $y = (\delta \overline{\tau}_{CH_4} - \delta \overline{\tau}_{IS})/\delta \overline{\tau}_{CH_4}$. A single mean value for each DAOD time series over the entire altitude leg is used to increase the accuracy of each DAOD estimate; the average leg duration was < 5 min. A relationship between y and $\delta \overline{\tau}_{CH_4}$ for each altitude leg is then represented by a cubic polynomial model, $y = \beta_0 + \beta_1 \delta \overline{\tau}_{CH_4} + \beta_2 \delta \overline{\tau}_{CH_4}^2 + \beta_3 \delta \overline{\tau}_{CH_4}^3$. A vector is then composed of the polynomial model for the entire maneuver, $\mathbf{y} = \mathbf{T}\boldsymbol{\beta}$, where \mathbf{y} is the vector of fractional differences, $\mathbf{T} = \begin{bmatrix} 1 & \delta \overline{\tau}_{CH_4} & \delta \overline{\tau}_{CH_4}^2 & \delta \overline{\tau}_{CH_4}^3 \\ \vdots & \vdots & \vdots \\ 1 & \delta \overline{\tau}_{CH_4} & \delta \overline{\tau}_{CH_4}^2 & \delta \overline{\tau}_{CH_4}^3 \end{bmatrix}$

is the matrix composed of $\delta \overline{\tau}_{CH_4}$, and β is the vector of biasdependent correction coefficients. A least-squares regression solves for β , which is then applied to correct the biased HALO DAOD as

$$\delta \tau_{\rm CH_4}' = \delta \tau_{\rm CH_4} \left[1 - \boldsymbol{\beta} \cdot \sum_{j=0}^3 \delta \tau_{\rm CH_4}^j \right].$$
⁽⁵⁾

This method is similar to that developed within Campbell et al. (2020) for altitude bias correction of CO₂ IPDA estimates.

Figure 6 shows an example of a four-level stair-step maneuver from the 11 June flight. HALO was operated in an "attenuated" mode for this calibration maneuver, seen in the DAOD time series as a constant standard deviation (SD) irrespective of flight altitude. The native $\delta \tau_{CH_4}$ for all receiver gains is shown in Fig. 6b with $\delta \tau_{IS}$ overlaid. Figure 6c shows the relationship between the native HALO DAOD and the computed fractional difference with respect to the in situ truth as a function of the fit parameters for each gain. In addition to differences in SNR and uncertainty in spectroscopy, fractional differences between HALO and in situ truth can result from differences in the differential transmission between the different optical channels and different differential transient responses between the different electrical gain channels. The absolute fractional difference is approximately 2 %-2.5 % for all altitudes, taken as the mean of all gains. Figure 6d shows the resulting data with the altitudedependent correction applied, indicating that the fitting routine yields a zero-bias relative to $\delta \overline{\tau}_{IS}$. The 1σ error bars in Fig. 6d represent the DAOD uncertainty per gain channel due to shot noise, indicating that the fitting routine will yield lower uncertainty for optimized receiver gains.

Results from the "unattenuated" mode of operation, where SNR increases for all channels as flight altitude decreases (contrasting a constant SNR with altitude in the "attenuated" mode), yielded comparable calibration results. The derived bias correction terms for the attenuated and unattenuated configurations were uniformly applied across all of the data collected throughout the mission in the attenuated and unattenuated modalities, respectively. Though each stair-step maneuver generates only a few data points in altitude for fitting, favorable comparisons of bias-corrected HALO DAOD with in situ observations throughout the campaign, as shown in subsequent sections (Figs. 10 and 11), demonstrate that the instrument calibration was stable over the duration of the mission. Furthermore, this indicates the presented correction method offers an interim solution to the observed biases while discrepancies in spectroscopy are investigated.

3.1.2 XCH₄ retrieval

The HALO observables used to retrieve the column XCH₄ are shown in Fig. 7. The data span a 50 km along-track flight segment for the low- and high-gain channels where the retrieval was optimized for the high gain. In each case the online and offline backscattered signals from the surface echo are digitized and summed on the FPGA. The integrated power from the surface echo is estimated at each wavelength, shown in Fig. 7a and e. The peak of the georeferenced ground return provides the SSE, shown in Fig. 7b and f in comparison to the DEM height. In this example, the SSE tracks the DEM closely; however, the optimized detection bandwidth and oversampling of the surface echo reveals the structure of the forest canopy. The integrated ground return is combined with the LEM measurement of pulse energies to calculate the DAOD according to Eq. (3) and bias corrected with Eq. (5), shown in Fig. 7c and g. Finally, the DAOD and weighting function are combined according to Eq. (4), to retrieve XCH₄, shown in Fig. 7d and h. The aircraft's GPS coordinate system is used for all calculations, and no additional steps are needed to align the 1645 nm backscatter to the DEM or MERRA-2 products.

The contrast in precision between gain channels in Fig. 7 is indicative of the SNR dependency of the XCH₄ retrieval and offers the ability to optimize the retrievals over a large dynamic range. The 30–40 km along-track portion of the high-gain column XCH₄ from Fig. 7h is further examined in Fig. 8, where histograms of the 2 Hz retrieved data from the optimized high-gain channel are shown against a 15 s averaging window for comparison. The 1 σ SD along this section gives 19.825 and 8.257 ppb and indicates a high preci-



Figure 6. Summary of the 11 June stair-step maneuver in eastern Virginia (VA) and steps to calculate a DAOD correction. (a) Flight profile and digital elevation model height. (b) The 2 Hz HALO DAOD for all gains and the in situ-derived DAOD. The transmit pulse energies are shown indicating variable attenuation to maintain a constant surface signal amplitude. (c) Fractional differences between the mean HALO and in situ-derived DAODs for the different gains. (d) Bias-corrected HALO DAOD compared to the in situ-derived values. Final 1σ SD values for each point are shown with in situ in black.

sion at short averaging scales. For HALO retrievals utilizing a gain channel that results in an optimal SNR, the HALO DAOD and resultant XCH₄ retrievals show comparable results for averaging scales similar to those previously published on CHARM-F (Amediek et al., 2017).

It was found that acceptable precision for all gain channels, ≤ 10 ppb, was reliably achieved with 15 s averaging windows. This was applied to all retrievals discussed here and was used to overcome noticeable decreases in precision experienced periodically throughout the campaign. To examine the retrieval precision the 1σ SD with different averaging times is computed, often described as the Allan deviation. Figure 9 shows an example of the noise statistics calculated from several flights across the central, southern, and eastern United States which exhibited varying surface structure, albedo, and flight altitude. Retrievals using a DAOD calculated from the non-optimized low-gain channel show a ~ 1 % SD (< 20 ppb) with < 10–15 s of averaging and ~ 0.5 %

(< 10 ppb) with 10–20 s of averaging. Retrievals made using optimized regions from the high-gain channel show a $\sim 1 \%$ SD with $\sim 1-5$ s of averaging and ~ 0.5 % with 5–10 s of averaging. Further averaging increases precision for applications that require high sensitivity, such as identifying weak emissions in thawing boreal regions. Although high precision can be achieved with relatively short averaging times, and different gains are employed to allow operational flexibility, the performance observed during ACT-America fell short of prior flights on the Langley B200 aircraft (Nehrir et al., 2018). The increased statistical noise observed could result from the harsh operating conditions on the C-130, resulting from slightly degraded laser frequency stability due to the high-vibration environment. Dedicated structural thermal and optical analysis of the laser transmitter subsystem was performed prior to full instrument test flights but did not indicate a significant degradation of performance (Fitzpatrick et al., 2019).



Figure 7. Example from the 11 June flight (3200 m a.g.l.) of the macroscopic processing steps involved in retrieving the column XCH₄ for a 50 km along-track segment at 2 Hz with the low-gain and high-gain channels shown on the left (**a**–**d**) and right (**e**–**i**), respectively. Panels (**a**) and (**e**) show the integrated ground return signals. Panels (**b**) and (**f**) show the calculated SSE and DEM. Panels (**c**) and (**g**) show the calculated DAOD reported at 2 Hz intervals. Panels (**d**) and (**h**) show the final retrieved XCH₄ at 2 Hz.



Figure 8. Histograms of the along-track XCH₄ retrievals, 30–40 km from Fig. 7h. Raw 2 Hz (0.5 s) data are shown in comparison to a 15 s average for the optimized high-gain signal. In each case the mean value and 1σ SD are shown.

Another contributing factor to the higher statistical uncertainty observed during ACT-America could result from speckle introduced by the long coherence length of the pulsed laser transmitter. HALO minimizes speckle within the receiver in two ways, first through the receiver by employing large collection apertures and secondly by employing alongtrack shot averaging, the latter of which will inherently break speckle cell correlation at the collection aperture on a shotby-shot basis. On the transmitter, the correlation of speckle cells must be broken between subsequent laser shots to measure the online / offline energy ratio of the transmitted laser pulses accurately, which is one of the main challenges of IPDA (Fix et al., 2018).

HALO's LEM employs a similar energy measurement method as reported in Fix et al. (2018). First, two integrating spheres are used to attenuate the sampled pulse to acceptable levels. A multi-mode optical fiber further attenuates the light circulating within the second integrating sphere and is used to transport the sampled pulse to the LEM detector. Diffusers are placed at the input aperture of the first and second integrating spheres and are used to break the correlation of speckle cells introduced by the rough surface of the integrating spheres themselves. The relatively small di-



Figure 9. XCH₄ noise statistics from the low gain (a) and high gain (b) for different terrains. The XCH₄ measurement precision at the native 2 Hz interval is $\sim 10-50$ ppb depending on terrain conditions and channel optimization. With a 15 s averaging window, ~ 2 km along-track, measurements approach a < 10 ppb precision, ~ 0.5 % assuming a 2000 ppb background. The along track distance assumes a 150 m s⁻¹ ground speed.

ameter of the collection fiber (105 µm) and slow oscillating frequency (180 Hz) of the speckle reducing diffusers, compared to the 1 kHz PRF of the pulsed laser, result in residual speckle cell correlation between the online and offline over several pulses. Zero-path calibration indicates that the speckle limited noise floor of the DAOD measurement is limited to ~ 0.005 over a half second average (250 shots per wavelength), where additional averaging provides further reduction. A recent MERLIN study (Cassé et al., 2019) showed that the impact of speckle on transmit energy measurements scales with SNR and that the expected random noise due to speckle for MERLIN approached $\leq 5 \text{ ppb}$ (or $\sim 0.25 \%$ for 2000 ppb) with < 10 s of averaging. These values are in line with HALO's findings and indicate the potential for speckle to dominate measurement noise if not accommodated for. Future investigations to further reduce speckle in HALO's LEM measurements are under investigation.

3.2 In situ validation

Vertical profiles of GHGs (CO_2 and CH_4 amongst others) and meteorological variables were periodically sampled in situ on each aircraft and offered a unique validation opportunity. An overpass of the in situ profile location prior to, or after, the C-130 spiral, descending or ascending, allowed for direct comparison of the lidar-derived XCH₄ to in situderived XCH₄.

An example of a spiral maneuver from the 20 July flight from ~ 5.2 –0.3 km a.g.l. and a ~ 12.5 km diameter overpass of the spiral is shown in Fig. 10. A 3D representation of the inbound and outbound flight line, overpass, and in situ CH₄ measurements is shown in Fig. 10a. The in situ CH₄ mole fraction profile is interpolated to HALO's vertical grid, shown in Fig. 10b in black, and is then used to derive an in situ XCH₄ retrieval from each flight altitude, shown in Fig. 10b in magenta. Comparing the CH₄ mole fraction profile to the in situ-derived XCH₄ in Fig. 10b facilitates an understanding of the differences between a point measurement at a given altitude and the equivalent columnweighted estimate from that altitude. In Fig. 10b, the highest in situ-derived XCH₄ retrieval (~ 5.2 km a.g.l.) provides the comparison value to HALO's estimate. The mean HALO XCH₄ retrieval from the overpass is also shown in Fig. 10b at 1.9086 ppm with an 9.46 ppb SD and compares to the in situ-derived XCH₄ estimate of 1.9001 ppm with an SD of $\pm < 1$ ppb. This gives a mean difference of 8.5 ppb, or 0.447 %, indicating that HALO has good agreement with the in situ measurement.

Each C-130 ascent or descent spiral profile that met requirements for lidar comparison (e.g., wing level, stabilized pulsed laser, low cloud extent) was used to evaluate HALO's XCH₄ retrievals. After screening, 11 of 23 spiral profiles (nine descent, two ascent) were used in comparison to HALO XCH₄ from the coincident overpasses. In some cases, spiral ascents were performed after long-duration boundary layer legs, resulting in an inability for the OPO to stabilize prior to the post-ascent overpass, while others had inadequate overpasses for HALO sampling. Some comparisons were carried out from a low flight altitude, which can limit lidar measurement precision (i.e., precision increases proportionally with DAOD). For each comparison a manual selection of the gain channel was used to optimize SNR. Figure 11 shows the correlation of the in situ XCH₄ from the spiral profiles to HALO's XCH₄ from the coincident overpasses. Each point is colored by the HALO DAOD and has a designation for spiral direction (ascent vs. descent). A correlation of R = 0.9058 was calculated for all comparisons, and we define the bias across all comparisons as the mean difference between HALO and the in situ-derived estimate, giving 2.54 ppb and a 1σ SD of the differences of 16.66 ppb. It should be noted that the comparison to in situ-derived CH₄ during stair-step maneuvers used for bias correction is not included within this comparison, and no additional calibrations were applied to the data collected throughout the campaign.

The locations of the spiral maneuvers analyzed in Fig. 11 were planned to be distant from strong local sources whenever possible. However, we expect that it is possible that a comparison could have unexpected enhancements below the minimum aircraft spiral altitude which are not represented within the in situ profile. Though unlikely, this could account for some of the differences seen between the two instruments. To understand such a scenario, and the subsequent impact on an in situ-derived column estimate, the profile in Fig. 10b is further examined. For a uniform 50 ppb enhancement added to the range bins from the lowest spiral altitude to the SSE (approximately 400 m), the in situ-derived column estimate changes by only \sim 4.5 ppb from 1.9004 to \sim 1.905 ppm, a 0.25 % increase. Interpreted through Fig. 11's results, the aggregate mean difference between HALO and in situ-derived XCH₄ decreases by < 1 ppb, indicating that this effect is likely not a major driver of the spread in random error. This does, however, emphasize the challenge in validation and evaluation methods for a column-integrating lidar, where enhancements not captured in situ, but seen by the lidar, would translate to changes of a few parts per billion over the total column and would be comparable with the total allowable systematic error, the example here accounting for one-half.

4 Regional-scale observations

ACT-America's regional sampling strategy and coordinated flights between the C-130 and B200 aircraft provided a unique opportunity to evaluate HALO's observations to in situ data over large regional scales. Near spatially coincident flight lines for C-130 and the B200 aircraft are shown in Fig. 12 from the 20 July flight. The spatial and temporal coordination between the two aircraft during this flight provided an ideal opportunity to assess the sensitivity of the HALO column XCH₄ measurements to variability within the PBL where surface fluxes dominate signals. Due to differing flight speeds, altitudes, and B200 refueling, the alignment of the two aircraft in time is offset until the latter portion of the flight, with the C-130 lagging the B200 by $\sim 2h$ at the start to the C-130 forward of the B200 by ~ 0.5 h at the end. HALO's XCH₄ and coincident HSRL aerosol backscatter are shown in Fig. 12a and b from the C-130 and in situ PBL CH₄ from the B200 in Fig. 12c. The associated HALO IPDA path length with the temporal separation of the two aircraft overlaid is shown in Fig. 12d. Screening of the B200 in situ measurements to the PBL utilized a combination of HALO's MLH and examination of the B200's in situ water vapor mixing ratio for transitions to the moist PBL, $\geq 14 \text{ g kg}^{-1}$. Figure 12a and c show good spatial agreement for the enhancements and magnitudes between HALO's column XCH₄ retrievals and the PBL in situ observations. Several regional enhancements (e.g., urban, agricultural, oil/gas) were observed by both instruments, and these spatially covarying signals provide qualitative indication that HALO's column XCH₄ has sensitivity to PBL CH₄ abundances.

Of particular interest is the S-N transect of the Pennsylvania (PA) region, where a significant enhancement is observed by both instruments. This broad enhancement is likely explained by emissions from the regional natural gas and coal production facilities (Barkley et al., 2019a). The transect is expanded in Fig. 13, where the time series of HALO XCH₄, in situ B200 PBL CH₄, and C-130 FT CH₄ are shown in Fig. 13a. At the lower latitudes of the transect HALO and the PBL in situ agree to within 25-50 ppb of each other $(\sim 1 \% - 2 \%$ difference), indicating that little to no enhancement is present within the lower troposphere and that the absolute magnitude of the column measurements correlates well with point measurements. A steady regional enhancement, maximizing at $\sim 150 \text{ ppb}$ above background, is seen by HALO and in situ from southern PA (40° N) to northern PA (42° N). Given the HALO weighting function, it is expected that HALO's measurement of the enhancement would be expected to be muted compared to the PBL in situ observations (like Fig. 10b). At the latter portion of the transect (north of $\sim 41.2^{\circ}$ N) the in situ enhancement subsides to background levels while HALO still measures a \sim 75– 100 ppb enhancement. These differences could arise if the FT air has elevated CH₄ originating from a different source than the more local emissions captured by the PBL observations. This hypothesis is supported by the appearance of an elevated aerosol layer in Fig. 12b that appears at approximately 40° N, the point where the HALO XCH₄ appears to increase with distance along the flight more rapidly than the in situ mole fractions (Fig. 13a). Closer examination of this layer in Fig. 13b shows that an inflow of air lofting aerosols into the FT is present, with the B200 in situ wind direction within the PBL indicating a southwesterly flow in the PA enhancement region. This elevated aerosol layer potentially originates from a PBL source far upwind of the flight line and thus may include elevated CH4 mole fractions. This could explain the divergence between the PBL and column CH4 measurements, particularly at the northern end of the flight track. These results show the sensitivity of XCH₄ measurements to advected enhancements, similar to the conclusions of Feng et al. (2019a, b) concerning XCO₂ observations. These comparisons demonstrate the value of HSRL in detecting these advected layers, the need for atmospheric transport models to interpret these data more fully, and the potential value of CH₄ profiling. Additional analysis with model comparisons, such as those conducted in Bell et al. (2020) for XCO₂, is required to definitively attribute the total column enhancement and will be the subject of future investigation.



Figure 10. (a) C-130 flight track and in situ CH₄ profile from the 20 July flight (\bigcirc Google Maps). (b) In situ CH₄ mole fraction profile as measured during a descent spiral from approximately 5 km to the ground height (SSE), in black. Overlaid is the in situ-derived XCH₄ in magenta using HALO's weighting function. The in situ-derived XCH₄ at flight altitude was $\mu_{IS} = 19001$ ppm, shown as the top point of the magenta curve, and the black error bars show HALO's overpass mean value with $\mu_{H} = 19086$ ppm and $1\sigma_{H} = 946$ ppb.



Figure 11. Comparison of the in situ-derived XCH₄ to HALO XCH₄ for 11 spirals, color coded by the HALO one-way DAOD. A correlation between in situ-derived and HALO XCH₄ gives R = 0.9058, with the fit shown as a red dashed line against the black one-to-one line. The 1% and 2% error bounds are shown. The 1 σ error bars are shown for the HALO XCH₄ retrievals.

To further assess the ability of a lidar column measurement to observe variability from near-surface emissions, the HALO XCH₄ was correlated to the PBL in situ observations from the B200 for the full flight, as in Fig. 12, as well as the PA S-N transect, as in Fig. 13. The comparisons were limited to 0.1° (~ 8 km) radial search between the HALO total column and B200 PBL data. Given a planned flight line overlap, this filter ensures the nearest latitude-longitude of each aircraft is used for comparison. The spatially filtered data yield a correlation of R = 0.3507 for the full flight and R = 0.4003for the S-N transect. Apart from the PA S-N transect, HALO underestimates the B200 observations on a whole but still captures the variability seen by the B200; however, this is expected as column averages exhibit influence from lower background values out of the PBL. Despite the relatively variable time separation and mismatch of sampling volumes between the two datasets, the correlation coefficients indicate mild correlation is present and further demonstrate the ability of column-integrating measurements to observe PBL variability.

A second example of a comprehensive dataset for comparison was collected on the 27 June flight in the southern portion of the Mississippi River valley. The comparison was divided into two comparison regions, XCH₄ vs. C-130 PBL CH₄ and XCH₄ vs. B200 and C-130 PBL CH₄. The first



Figure 12. Comparisons between HALO's column XCH₄ from the C-130 and in situ PBL CH₄ from the B200 during the second leg of the 20 July flight. (a) Cloud-cleared HALO XCH₄ retrievals (© Google Maps). (b) HSRL-derived aerosol backscatter at 532 nm with an overlay of the TERRA MODIS corrected reflectance to indicate cloud extent (© Google Maps). (c) In situ sampled CH₄ restricted to the PBL (the B200 aircraft landed for refueling, ~ 18.5 –19.25) (© Google Maps). (d) IPDA path length with the C-130 and B200 temporal separation.

comprised the western leg where the C-130 flew at altitude $(\sim 6.5 \text{ km})$ to collect XCH₄ and then in the reverse direction within the PBL (< 1 km a.g.l.) to sample the same background region in situ. The second comprised the southern legs where the C-130 flew in the FT to make in situ and XCH₄ measurements in coordination with the B200 sampling CH₄ in the PBL. The C-130 subsequently flew within the PBL for in situ sampling on a northern return to provide an indication of how the PBL enhancement changed spatially. Figure 14a shows all regions of coincident HALO XCH₄ and PBL in situ CH₄, from each aircraft for both regions. Here, HALO is at altitude within the FT, and the C-130 and B200 legs are within the PBL. Multiple regions show covariance between the remotely sensed column and the in situ PBL observations. Figure 14b shows the lower central region where the enhancement comparisons take place. Here all instruments register enhancements, emphasized within the lower flight track section of Fig. 14b. Given the location of these flight lines, it is likely that these plumes are indicative of wetland emissions. To provide further context, a curtain of the HSRL aerosol backscatter with the overlaid PBLH in red is shown in Fig. 14c. Signals attenuated beneath opaque clouds are masked out in black and provide insight into the atmospheric state during the sampling time. In the

earlier portion of the day the background comparison region portion of the flight exhibits a shallower PBL (pre-noon local standard time), whereas in the latter portion of the day where the enhancement comparisons occur a deeper PBL has developed, and significant aerosol lofting has occurred. The PBL wind direction measured in situ by the C-130 and B200 indicates a complicated wind scene with a general NW flow in the enhancement region and low wind speeds of 5–10 kn.

The correlation analysis was applied to the two comparison regions from Fig. 14a and b. The stacked out and back legs on the background comparison region exhibit near-zero correlation at R = 0.0792, indicating that column measurements made in background conditions which are void of large emissions do not correlate with weaker surface fluxes and that elevated signal captured by HALO was not sampled in situ during the PBL return leg. The high- and lowaltitude flight lines shown in Fig. 14b, sampled by the remote and in situ instruments respectively, demonstrate a higher degree of correlation at R = 0.7218 between the B200 PBL measurement and HALO and R = 0.4290 between the C-130 PBL measurement and HALO. The combined correlation analysis of HALO to both PBL in situ instruments in the enhancement region exhibits a correlation of R = 0.6075. Examining Fig. 14b, the measurements indicate a delineation



Figure 13. South–north transect, $39-42.3^{\circ}$ N, of Pennsylvania from Fig. 12. (a) Mole fractions measured by each instrument, HALO column XCH₄ in red, the B200 PBL in situ CH₄ in black, and C-130 in situ FT CH₄ in blue. (b) Cloud-cleared (black vertical lines) HSRL aerosol backscatter at 532 nm during the transect and DEM height, white dots. (c) IPDA path length. (d) Altitude above ground level (a.g.l.) time series of each aircraft.

between background and the enhancement, and despite a difference in absolute magnitude the spatially defined enhancements captured by all instruments provide further indication that column-derived XCH₄ measurements can be used as an indicator for PBL enhancement.

5 Advanced CH₄ products – atmospheric profiling

The DIAL technique uses ratios of atmospheric signals to derive a relative DAOD and the number density within a prescribed range interval. Using atmospheric signals directly, DIAL is self-calibrating and overcomes many of the challenges associated with IPDA to generate a column measure-



Figure 14. Overlapping flight lines between the C-130 and B200 aircraft in the southern Mississippi River valley region from the 27 June flight. (a) Combined flight lines for low altitude PBL in situ observations from C-130 and B200 and HALO XCH₄ from the C-130 at high altitude ($^{\odot}$ Google Maps). (b) Strong regional correlation can be seen between all three instruments, HALO at high altitude and the C-130 and B200 PBL observations at low altitude ($^{\odot}$ Google Maps). Prevailing winds measured in situ within the PBL indicate a NW flow during the observation period. (c) HSRL aerosol backscatter for the high-altitude C-130 legs with the PBLH overlaid.

ment (zero-path calibration, bias correction, and reference energy measurement). Benefits of higher precision are also afforded with DIAL as the retrieval is non-linearly proportional to the range bin size (Nehrir et al., 2017; Carroll et al., 2022) such that the large vertical averages required to increase the per bin number of photons will also increase the CH₄ DAOD precision (SNR values in excess of 500 are required for highly precise DIAL–IPDA retrievals). Although absolute knowledge of the total DAOD is not needed for a typical DIAL retrieval, here we have chosen to normalize the backscattered signals throughout the profile to near-aircraft signals to compare the atmospherically derived cumulative DAOD to the IPDA column DAOD.

Coincident measurements of the range-corrected offline backscattered signal and the HSRL 532 nm aerosol backscatter for the duration of the 20 July flight are shown in Fig. 15, where the offline backscatter was averaged to 15 m vertical resolution and 10 s along track to match the HSRL retrieval resolution. The two data curtains qualitatively demonstrate the ability of the DIAL–IPDA channels to capture key atmospheric features needed to enhance IPDA column measurements with profiling capabilities. Figure 15c shows the vertical profiles of HSRL aerosol and offline backscatter collected over the spiral overpass region analyzed in Fig. 10. The profiles indicate that the offline SNR is sufficient for a range-resolved retrieval; however retrieval quality and effectiveness are limited by the online wavelength's optical depth. This is further examined within Fig. 16.

To perform a range-resolved DIAL retrieval and estimate profiles of DAOD from the 1645 nm online and offline backscattered signals, a modified version of Eq. (2) is used as

$$\delta \tau_{\rm CH_4}^{\rm DIAL}(R) = \frac{1}{2} \ln \left(\frac{P_{\rm rx}\left(\lambda_{\rm off}, R\right) / P_{\rm rx}\left(\lambda_{\rm off}, R_{\rm norm}\right)}{P_{\rm rx}\left(\lambda_{\rm on}, R\right) / P_{\rm rx}\left(\lambda_{on}, R_{\rm norm}\right)} \right).$$
(6)

Unlike the IPDA-derived DAOD, the range-resolved calculation utilizes backscatter profiles which have been normalized by atmospheric signal from the nearfield of the aircraft, $R_{\rm norm}$. The normalization signal's altitude is chosen such that full geometric overlap has been achieved while also ensuring that appreciable CH₄ DAOD has not accumulated in the bins. This provides a comparable method to estimating the cumulative DAOD over the lidar profile for comparison to traditional IPDA estimates and without an ancillary LEM module to characterize the difference in online and offline pulse energies. In practice, the best placement of R_{norm} could still yield non-negligible amounts of CH₄ DAOD between the aircraft and the normalization point, < 0.01 for the comparisons during ACT-America. For a robust comparison to IPDA, this additional optical depth must be estimated and included within the cumulative estimate per range bin. When present, this is estimated by calculating the DAOD difference between the nearest signal to the aircraft and the normalization bin as $\delta \tau_{CH_4}^{DIAL}(R_{norm}) - \delta \tau_{CH_4}^{DIAL}(R_a)$, which is then added to each bin of the DAOD profile. The benefit of a range-resolved DAOD profile calculated with Eq. (6) is that no bias correction is applied, and the energy differences between pulses are measured within the atmospheric profiles.

To investigate CH₄ profiling capabilities, the spiral overpass presented in Figs. 10 and 15 was further examined, allowing simultaneous comparison of the in situ-derived DAOD, HALO IPDA-derived DAOD, and HALO DIALderived DAOD. Figure 16a and b show subsections of the range-corrected offline and online backscatter centered around the spiral location, where differential absorption between the DIAL-IPDA wavelengths can clearly be seen within the PBL backscatter. Figure 16c shows $\delta \tau_{CH_4}^{DIAL}(R)$ calculated with Eq. (6) for the duration of the overpass. The online and offline backscatter profiles were first averaged 15 s along track (2 km) and then to 350 m in the vertical prior to use in Eq. (6). Increasing DAOD can be seen from the FT into the PBL with an average value in the lowest retrieved bin approaching ~ 0.275 (one-way DAOD). Additional features can be seen within the DAOD curtain that correlate with the aerosol field, such as the clear air feature at ~ 2.5 km at the latter section of the overpass. This feature appears to be a manifestation of noisy low-SNR retrievals made in this region, resulting from low aerosol backscatter and larger standoff distance to the aircraft, and not the result of decreased CH₄ optical depth. The impact of low SNR can manifest as a statistically induced DAOD bias within the associated retrieval bin when utilizing Eq. (6). This effect has been well documented within Gibert et al. (2006, 2008), which indicates that the magnitude of the bias can be considered negligible for high-SNR backscatter (SNR > 10) that has aggregated signal over multiple shots and or multiple range bins (Gibert et al., 2006). It was found that the averaged profiles of backscatter exhibit high SNR throughout the majority of the averaged profile. Examining Fig. 17b, the online wavelength's SNR at the top of the PBL is ~ 60 , and the SNR at the low backscatter feature at ~ 2.4 km is ~ 5 . This gives an indication that the regions of interest near the surface and within the PBL exhibit higher precision due to the higher per-bin SNR for each wavelength.

The online and offline backscatter signals were further averaged over the entire overpass window to 350 m by 12.5 km to increase SNR and precision, giving a single range-resolved retrieval for the entire overpass. Figure 17a shows the input backscatter profiles, and the DAOD profile is shown in Fig. 17b. Here the near linear trend in the lower tropospheric DAOD is fully observed and is the result of the uniform weighting of absorption due to pressure broadening of the line complex in the lower atmosphere. The inset in Fig. 17b shows the DIAL- and IPDA-derived column estimates along with the in situ-derived DAOD from the overpass' spiral. Due to the required vertical averaging for the DIAL retrieval, the last atmospheric bin above the SSE is unresolved, setting the accumulated DAOD in the lowest retrieved atmospheric bin at 0.2723. To provide a comparable estimate to the IPDA-derived value at the SSE, a linear re-



Figure 15. Examination of 1645 nm offline backscatter from 20 July. (a) Range-corrected offline backscatter profiles (15 m; 10 s). (b) HSRL aerosol backscatter at 532 nm (15 m; 10 s) with MLH in white. The flight track, magenta, and DEM height, white, are shown in each panel. Each curtain was cloud cleared with the HSRL cloud top height, black striations. (c) Profiles of offline backscatter and the HSRL aerosol backscatter from the vertical white lines shown in panels (a)–(b) (15 m; 12.5 km).

gression was performed on the DIAL-calculated profile and extrapolated to the SSE, shown in Fig. 17b, giving an estimate of $\delta \tau_{CH_4}^{DIAL}(R_{SSE}) = 0.2943$. This contrasts the IPDA and in situ estimates of $\delta \tau_{CH_4} = 0.2837$ and $\delta \tau_{IS} = 0.2829$ at R_{SSE} and indicates that the DIAL-derived DAOD overestimates the total column estimates, IPDA and in situ, by 3.66 % and 3.95 %, respectively. The magnitude of the differences between the two independent measurements is on the order of the differences between the non-bias-corrected IPDA DAOD and the in situ-derived DAOD shown in Fig. 6 and provides further insight into the uncertainties associated with the CH₄ line parameters/spectroscopy used in the derivation of in situ-derived XCH₄ and within HALO XCH₄ retrievals.

5.1 Planetary boundary layer apportionment

Traditional methods for apportioning the PBL mole fraction from IPDA column measurements have relied on the "cloud slicing" technique (Ramanathan et al., 2015; Amediek et al., 2017). This method requires that fair weather cumulus and stratocumulus clouds cap the PBL and that IPDA columns measured to surface and cloud top can be subtracted and used to infer abundances of GHGs within the PBL, $\delta \tau_{PBL}^{IPDA} = \delta \tau_{ground}^{IPDA} - \delta \tau_{cloud}^{IPDA}$. Though this method has shown utility in retrieving near surface mole fractions, its usability diminishes in regions and conditions void of clouds. Figure 18a shows the time series of IPDA DAOD surrounding the overpass in Fig. 16, where fair weather cumulus clouds at PBL top prior to the overpass provide lower DAOD estimates and changes in SSE translate directly to changes in DAOD. Histograms for the entire window are shown in Fig. 18b, binned in DAOD increments of 0.001, where the distributions of DAOD at cloud top and ground are clearly delineated and enable an estimate of $\delta \tau_{\text{PBL}}^{\text{IPDA}}$ using the cloud slicing method. From the histograms, mean values from each DAOD dis-tribution were estimated as $\delta \tau_{\text{ground}}^{\text{IPDA}} = 0.2848$ and $\delta \tau_{\text{cloud}}^{\text{IPDA}} =$ 0.2164 and the subsequent PBL DAOD of $\delta \tau_{\text{PBL}}^{\text{IPDA}} = 0.0683$. Though this shows the ease at which $\delta \tau_{\text{PBL}}$ can be computed directly from the DAOD time series when clouds are present, the methods are restricted to the presence of clouds, and findings are extrapolated to the clear-air overpass. Given that the overpass region is mostly cloud free, the prior clouds provide the information required for comparison between $\delta \tau_{\text{PBL}}$ estimates derived from cloud slicing, and the clear- air overpass region can be computed through a DIAL. Combined, both methods bring about the potential for complementing measurements in variable atmospheric states and allow a contiguous measurement throughout cloudy and cloud-free regions.

For the clear-air region, $\delta \tau_{CH_4}^{DIAL}(R)$ can be used to estimate the DAOD PBL top, which can then be subtracted from the total-column DAOD to give the relative PBL contri-



Figure 16. The 1645 nm range-corrected backscatter (15 m; 10 s) from the 20 July in situ overpass subsection of Fig. 15. (a) Offline, 1645.3724 nm. (b) Online, 1645.5518 nm. The white bars in each panel indicate the overpass of the spiral location; see Fig. 10. Each curtain was cloud cleared and the DEM is overlaid. Panel (c) shows the calculated range-resolved DAOD at 350 m vertical and 15 s of along-track averaging within the overpass region defined in panels (a) and (b). The lowest retrieval bin occurs one 350 m range cell above the DEM due to the large vertical retrieval window.



Figure 17. (a) The 1645 nm range-corrected online and offline over the overpass region averaged to 350 m vertical by 12.5 km along-track resolution. (b) The range-resolved DAOD as the black line, with fitted DAOD from the range-resolved profile shown by the magenta dashed line. The magenta "diamond" emphasizes the fitted value at the SSE, the IPDA-derived DAOD is the black circle with 1σ error bars, and in situ-derived DAOD is the red box.



Figure 18. Summary of PBL DAOD derivation for the overpass shown in Fig. 16. (a) The column and cloud top DAOD surrounding the overpass region with DIAL-derived and in situ DAOD estimates of shown. (b) Average DAOD from cloud top and to the SSE, where histograms indicate the peak estimates from which the PBL abundances can be derived using cloud slicing, giving $\delta \tau_{PBL}^{IPDA} = 0.0683$ for the duration of the window. (c) Compares the IPDA SSE ground and cloud top height to the DEM.

bution as $\delta \tau_{\text{PBL}}^{\text{DIAL}} = \delta \tau_{\text{CH}_4}^{\text{DIAL}} (R_{\text{SSE}}) - \delta \tau_{\text{CH}_4}^{\text{DIAL}} (R_{\text{MLH}})$. Given HALO's measurement modalities, the HSRL-derived MLH can be used to discern the PBL top and the IPDA SSE can be used for the ground elevation, indicated in Fig. 19. From the DAOD fit the DAOD at each altitude can be extracted to give a DAOD estimate for the PBL column as $\delta \tau_{\text{PBL}}^{\text{DIAL}} = 0.0557$. Comparing to the in situ-derived DAOD for the portion of the column, $\delta \tau_{\text{PBL}}^{\text{IS}} = 0.0561$ was estimated from the spiral profile when using HALO's HSRL MLH and IPDA SSE as integration bounds. The relative components for the PBL column from each computation are shown within Fig. 18 in contrast to the cloud slicing estimate. Utilizing the in situ temperature and pressure profiles from the spiral, a subset of the HALO weighting function for the PBL was used to derive a PBL column mole fraction of 1.9629 ppm, from $\delta \tau_{\text{PBL}}^{\text{DIAL}}$, and 1.9775 ppm, for $\delta \tau_{\text{PBL}}^{\text{IS}}$. This gives a difference of ~ 0.741 % and indicates that the HALO DIAL method has the potential to provide clear-air estimates of PBL XCH₄.

to provide clear-air estimates of PBL XCH₄. Further examination of $\delta \tau_{PBL}^{DIAL}$ and $\delta \tau_{PBL}^{IS}$ indicates that they differ from the estimate derived using cloud slicing, the latter of which appears to provide an absorption overestimate when extended to the clear-air region. This is likely due to differing mole fractions between the air masses such that extrapolation is not valid or due to spectroscopy-induced error resulting from application of the correction described in Sect. 3.1.1 to the base IPDA retrievals used within the cloud slicing computation. Figure 18 indicates that absorption estimates to PBL top and over the total column are consistent for all three methods, despite exhibiting minor differences. When estimating PBL-specific absorption however, small PBL DAOD uncertainties, even on the order of 0.001– 0.002, can translate to several percent of uncertainty in the derived geophysical observable XCH₄.

Utilization of this method for future PBL-focused studies requires further development to document uncertainties from the DIAL retrieval. As shown here, the DIAL retrieval suffers from lower SNR compared to the IPDA retrievals, which benefit from the strong surface returns. For the cases where high DIAL SNR can be achieved with moderate alongtrack averages (requiring increased PBL backscatter or significantly improved detection methods), this retrieval could provide new insights into PBL mole fractions in clear-air regimes. To complement the DIAL retrieval and extend measurements down to the surface without the need for linear fitting and extrapolation, a hybrid-IPDA (HIPDA) method has been devised which utilizes the atmospheric signals at PBL top and the strong surface return to directly apportion the PBL DAOD from the column (e.g., filling in the 350 m above the SSE). The HIPDA method is similar to that employed in HALO's WV DIAL retrievals (Carroll et al., 2022), where the DAOD due to WV between the lowest retrieval bin and the SSE is estimated and used to extend the DIAL-derived mole fraction through the entire PBL. HIPDA is currently being adapted to the CH₄ retrieval; however, validation of



Figure 19. Range-resolved DAOD at 350 m vertical by 12.5 km along-track resolutions, black, with fitted estimate overlaid, grey, with extrapolation to the SSE. The HSRL MLH and IPDA SSE are shown for the overpass region and allow an estimate of $\delta \tau_{PBL}^{DIAL}$ from the fitted profile. The in situ CH₄ profile from the subsequent spiral profile, blue, is shown for comparison.

the technique has not been performed and will be the subject of a future publication.

6 Conclusion

The HALO CH₄ DIAL–IPDA measurements were quantitatively evaluated for the first time during the 2019 ACT-America campaign. Data were collected from the NASA C-130 aircraft during 18 of the 19 flights and two engineering flights. These flights were the first detailed validation efforts of a combined CH₄ DIAL–IPDA and HSRL, demonstrating a unique ability to contextualize CH₄ column measurements with additional information afforded by the HSRL and backscatter profiles. Data collected during this mission provided a unique opportunity for validation and assessment of instrument stability and retrieval accuracy and precision. Additionally, the data provided insight into future investigations, such as optimization of spectroscopic line parameters which currently serve as the largest source of uncertainty in the HALO XCH₄ retrieval.

Analysis across the duration of the campaign found that the single point calibration of HALO's CH₄ channels coupled with the overall stability of the HALO instrument provided repeatable and reliable measurements of XCH₄ over a wide range of atmospheric and surface conditions aboard an environmentally challenging aircraft. Data collected over varying terrain were used to compute noise statistics for the high- and low-gain channels and showed that a precision of 0.5% was achievable for averaging intervals of < 15 s in the low-gain channel and < 10 s in the high-gain channel, allowing for operation at different aircraft altitudes and over different surface albedos. Comparisons of HALO to in situ-derived column estimates were carried out throughout the campaign, where in situ profiles were generated during spiral ascents or descents under the overpass region and provided validation of HALO's XCH₄ measurements. An overall correlation of R = 0.9058 with a bias across all comparisons, the mean difference between HALO and the in situ-derived estimate of 2.54 ppb, and a 1σ SD of the differences of 16.66 ppb across all 11 comparisons were observed. Given HALO operated in vastly different research modes to optimize for emerging atmospheric profiling CH₄ retrievals, an improvement in reducing the required along-track averaging to achieve consistent < 1% precision is expected in future flights. This can be achieved by optimizing the transmit energy (or the receiver optical splits between different gain channels) to better utilize the high optical signals for the IPDA measurement. Lastly, several comparisons of lidar-derived XCH₄ and in situ measurements of CH₄ within the PBL were made at regional scales and showed high degrees of covariance. These demonstrated the ability of a column-integrating lidar to observe CH₄ variability within the PBL where CH₄ fluxes dominate signals.

An altitude-dependent bias of < 2.5 % (average) was identified in HALO's DAOD when compared to in situ-derived DAOD. These biases were removed by correcting the lidar measurements to in situ truth through a stair-step maneuver carried out in background conditions assumed void of known enhancements. A single set of corrections was applied to each channel for the entirety of the campaign. The resulting bias-corrected data showed excellent agreement with in situ spiral profiles for the campaign duration, demonstrating the instrument stability and validating the correction method employed. The bias source has been investigated, and all indicators point towards an uncertainty in the spectroscopic line parameters derived from HITRAN 2016. The impact of statistical biases induced by low-SNR retrievals (Tellier et al., 2018) was investigated and found to impact retrievals made with a non-optimized receiver gain, indicating that correct selection of a gain channel relegated this bias source as negligible. Our initial spectroscopic bias conclusions also agree with findings published in preparation for the MERLIN mission (Delahaye et al., 2016b; Vasilchenko et al., 2019) as well as those found by the CHARM-F IPDA lidar instrument (Fix et al., 2020). Future work will incorporate updated spectroscopy and statistical/geophysical corrections into the XCH₄ retrievals, and a bias reduction/removal is anticipated.

During the 2019 ACT-America flights HALO demonstrated, for the first time in a scientific setting, range-resolved measurements of CH₄ DAOD employing the DIAL technique. The DIAL technique can overcome the primary challenge associated with IPDA, namely the requirement of accurate knowledge of the transmitted energy ratio and receivertransmit path differential transmission ratio, which serves as the two largest sources of uncertainty in an IPDA lidar. Longer horizontal averages than typically utilized for IPDA, \sim 12 km, were employed to increase the DIAL retrieval SNR, a result of weakly scattering atmospheric aerosols and molecules compared to the strong surface signal. The DIALderived DAOD at the SSE was compared to the standard IPDA and the in situ-derived estimates, showing good agreement with < 1% retrieval accuracy. We expand further on these atmospheric retrievals by demonstrating the novel ability to directly apportion the PBL DAOD from the column in clear-air conditions using the range-resolved DAOD profiles. Comparisons of the HALO-derived PBL DAOD/XCH₄ to the in situ-derived PBL column showed favorable agreement, on the order of 1 % absolute difference, and provide a foundation of understanding needed to make CH₄ atmospheric profiling an operational product for future campaigns. To enable this, future instrument enhancements include the use of higher-sensitivity HgCdTe detectors and further optimized gain settings between the DIAL and IPDA channels. The range-resolved DIAL methods presented herein have the potential to provide new insights into CH₄ fluxes across scales and offer an avenue for the first remotely sensed profiles of atmospheric CH₄ with the needed sensitivity for inventory and survey studies. The added HSRL observations made by HALO also provide unique contextual information that will be critical for validation of future passive CH4 measurements from space.

Data availability. ACT-America observational and modeling datasets are archived at the ORNL DAAC (https://doi.org/10.3334/ORNLDAAC/1556, DiGangi et al., 2018; https://doi.org/10.3334/ORNLDAAC/1862, Pal and Davis, 2021). HALO CH_4 products are not yet available at the ORNL DAAC but are available upon request.

Author contributions. RABG led the analysis presented here with contributions from ARN. SK, JC, and ARN led data curation and development of the HALO HSRL products. RABG, SAK, ARN, and JEC developed HALO's methane retrievals. RABG, ARN, JEC, DBH, and JL contributed to the preparation and deployment of the HALO instrument. JPD and YC provided in situ data collected during the campaign for comparison to HALO. KJD led the ACT-America campaign, designed the flights, and directed execution of the flight patterns. RABG prepared the manuscript with contributions from co-authors.

Competing interests. The contact author has declared that none of the authors has any competing interests.

Disclaimer. Publisher's note: Copernicus Publications remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Acknowledgements. We acknowledge funding support from the NASA Headquarters Earth Science Division, the NASA Earth Science Technology Office, and the NASA Langley Research Center. We thank the C-130 and B200 teams at the NASA Wallops Flight Facility and NASA Langley Research Center, respectively, and the National Suborbital Education and Research Center for their support of the ACT-America campaign. We acknowledge the use of imagery from the NASA Worldview application (https://worldview.earthdata.nasa.gov/, last access: 22 July 2022), part of the NASA Earth Observing System Data and Information System (EOSDIS). The Atmospheric Carbon and Transport (ACT)-America project is a NASA Earth Venture Suborbital-2 project funded by NASA's Earth Science Division (grant NNX15AG76G to Penn State).

Financial support. The Atmospheric Carbon and Transport (ACT)-America project is a NASA Earth Venture Suborbital-2 project funded by NASA's Earth Science Division (grant no. NNX15AG76G to Penn State).

Review statement. This paper was edited by Christoph Kiemle and reviewed by two anonymous referees.

References

- Abshire, J. B., Ramanathan, A. K., Riris, H., Allan, G. R., Sun, X., Hasselbrack, W. E., Mao, J., Wu, S., Chen, J., Numata, K., Kawa, S. R., Yang, M. Y. M., and DiGangi, J.: Airborne measurements of CO₂ column concentrations made with a pulsed IPDA lidar using a multiple-wavelength-locked laser and HgCdTe APD detector, Atmos. Meas. Tech., 11, 2001–2025, https://doi.org/10.5194/amt-11-2001-2018, 2018.
- Alvarez, R. A., Zavala-Araiza, D., Lyon, D. R., Allen, D. T., Barkley, Z. R., Brandt, A. R., Davis, K. J., Herndon, S. C., Jacob, D. J., Karion, A., and Kort, E. A.: Assessment of methane emissions from the US oil and gas supply chain, Science, 361, 186–188, 2018.
- Amediek, A., Sun, X., and Abshire, J. B.: Analysis of Range Measurements From a Pulsed Airborne CO₂ Integrated Path Differential Absorption Lidar, IEEE T. Geosci. Remote, 51, 2498– 2504, 2013.
- Amediek, A., Ehret, G., Fix, A., Wirth, M., Büdenbender, C., Quatrevalet, M., Kiemle, C., and Gerbig, C.: CHARM-F – a new airborne integrated-path differential-absorption lidar for carbon dioxide and methane observations: measurement performance and quantification of strong point source emissions, Appl. Optics, 56, 5182–5197, 2017.
- Amediek, A., Fix, A., Ehret, G., Caron, J., and Durand, Y.: Airborne lidar reflectance measurements at 1.57 μm in support of the A-SCOPE mission for atmospheric CO₂, Atmos. Meas. Tech., 2, 755–772, https://doi.org/10.5194/amt-2-755-2009, 2009.

- Andrews, A. E., Kofler, J. D., Trudeau, M. E., Williams, J. C., Neff, D. H., Masarie, K. A., Chao, D. Y., Kitzis, D. R., Novelli, P. C., Zhao, C. L., Dlugokencky, E. J., Lang, P. M., Crotwell, M. J., Fischer, M. L., Parker, M. J., Lee, J. T., Baumann, D. D., Desai, A. R., Stanier, C. O., De Wekker, S. F. J., Wolfe, D. E., Munger, J. W., and Tans, P. P.: CO₂, CO, and CH₄ measurements from tall towers in the NOAA Earth System Research Laboratory's Global Greenhouse Gas Reference Network: instrumentation, uncertainty analysis, and recommendations for future high-accuracy greenhouse gas monitoring efforts, Atmos. Meas. Tech., 7, 647–687, https://doi.org/10.5194/amt-7-647-2014, 2014.
- Barkley, Z. R., Lauvaux, T., Davis, K. J., Fried, A., Weibring, P., Richter, D., Walega, J. G., DiGangi, J., Ehrman, S. H., Ren, X., and Dickerson, R. R.: Estimating methane emissions from underground coal and natural gas production in southwestern Pennsylvania, Geophys. Res. Lett., 46, 4531–4540, 2019a.
- Barkley, Z. R., Davis, K. J., Feng, S., Balashov, N., Fried, A., Di-Gangi, J., Choi, Y., and Halliday, H. S.: Forward Modeling and Optimization of Methane Emissions in the South Central United States Using Aircraft Transects Across Frontal Boundaries, Geophys. Res. Lett., 46, 13564–13573, 2019b.
- Barkley, Z. R., Davis, K. J., Feng, S., Cui, Y. Y., Fried, A., Weibring, P., Richter, D., Walega, J. G., Miller, S. M., Eckl, M., Roiger, A., Fiehn, A., and Kostinek, J.: Analysis of oil and gas ethane and methane emissions in the southcentral and eastern United States using four seasons of continuous aircraft ethane measurements, J. Geophys. Res.-Atmos., 126, e2020JD034194, https://doi.org/10.1029/2020JD034194, 2021.
- Bedka, K. M., Nehrir, A. R., Kavaya, M., Barton-Grimley, R., Beaubien, M., Carroll, B., Collins, J., Cooney, J., Emmitt, G. D., Greco, S., Kooi, S., Lee, T., Liu, Z., Rodier, S., and Skofronick-Jackson, G.: Airborne lidar observations of wind, water vapor, and aerosol profiles during the NASA Aeolus calibration and validation (Cal/Val) test flight campaign, Atmos. Meas. Tech., 14, 4305–4334, https://doi.org/10.5194/amt-14-4305-2021, 2021.
- Bell, E., O'Dell, C. W., Davis, K. J., Campbell, J., Browell, E., Scott Denning, A., Dobler, J., Erxleben, W., Fan, T. F., Kooi, S., and Lin, B.: Evaluation of OCO-2 X variability at local and synoptic scales using lidar and in situ observations from the ACT-America campaigns, J. Geophys. Res.-Atmos., 125, e2019JD031400, https://doi.org/10.1029/2019JD031400, 2020.
- Bousquet, P., Ciais, P., Miller, J. B., Dlugokencky, E. J., Hauglustaine, D. A., Prigent, C., Van der Werf, G. R., Peylin, P., Brunke, E. G., Carouge, C., and Langenfelds, R. L.: Contribution of anthropogenic and natural sources to atmospheric methane variability, Nature, 443, 439–443, 2006.
- Bousquet, P., Ringeval, B., Pison, I., Dlugokencky, E. J., Brunke, E.-G., Carouge, C., Chevallier, F., Fortems-Cheiney, A., Frankenberg, C., Hauglustaine, D. A., Krummel, P. B., Langenfelds, R. L., Ramonet, M., Schmidt, M., Steele, L. P., Szopa, S., Yver, C., Viovy, N., and Ciais, P.: Source attribution of the changes in atmospheric methane for 2006–2008, Atmos. Chem. Phys., 11, 3689–3700, https://doi.org/10.5194/acp-11-3689-2011, 2011.
- Browell, E. V., Ismail, S., and Grant, W. B.: Differential absorption lidar (DIAL) measurements from air and space, Appl. Phys. B, 67, 399–410, 1998.
- Bruhwiler, L., Dlugokencky, E., Masarie, K., Ishizawa, M., Andrews, A., Miller, J., Sweeney, C., Tans, P., and Worthy, D.: CarbonTracker-CH₄: an assimilation system for estimating emis-

sions of atmospheric methane, Atmos. Chem. Phys., 14, 8269–8293, https://doi.org/10.5194/acp-14-8269-2014, 2014.

- Campbell, J. F., Lin, B., Dobler, J., Pal, S., Davis, K., Obland, M. D., Erxleben, W., McGregor, D., O'Dell, C., Bell, E., and Weir, B.: Field evaluation of column CO2 retrievals from intensitymodulated continuous-wave differential absorption lidar measurements during the ACT-America campaign, Earth Space Sci., 7, e2019EA000847, https://doi.org/10.1029/2019EA000847, 2020.
- Carroll, B. J., Nehrir, A. R., Kooi, S. A., Collins, J. E., Barton-Grimley, R. A., Notari, A., Harper, D. B., and Lee, J.: Differential absorption lidar measurements of water vapor by the High Altitude Lidar Observatory (HALO): retrieval framework and first results, Atmos. Meas. Tech., 15, 605–626, https://doi.org/10.5194/amt-15-605-2022, 2022.
- Cassé, V., Gibert, F., Edouart, D., Chomette, O., and Crevoisier, C.: Optical energy variability induced by speckle: The cases of MERLIN and CHARM-F IPDA lidar, Atmosphere, 10, 540, https://doi.org/10.3390/atmos10090540, 2019.
- Cui, Y. Y., Brioude, J., McKeen, S. A., Angevine, W. M., Kim, S. W., Frost, G. J., Ahmadov, R., Peischl, J., Bousserez, N., Liu, Z., and Ryerson, T. B.: Top-down estimate of methane emissions in California using a mesoscale inverse modeling technique: The South Coast Air Basin, J. Geophys. Res.-Atmos., 120, 6698–6711, 2015.
- Cusworth, D. H., Duren, R. M., Thorpe, A. K., Olson-Duvall, W., Heckler, J., Chapman, J. W., Eastwood, M. L., Helmlinger, M. C., Green, R. O., Asner, G. P., and Dennison, P. E.: Intermittency of large methane emitters in the Permian Basin, Environ. Sci. Tech. Let., 8, 567–573, 2021.
- Davis, K. J., Browell, E. V., Feng, S., Lauvaux, T., Obland, M. D., Pal, S., Baier, B. C., Baker, D. F., Baker, I. T., Barkley, Z. R., and Bowman, K. W.: The atmospheric carbon and transport (ACT)-America mission, B. Am. Meteorol. Soc., 102, E1714–E1734, 2021.
- Dean, J. F., Middelburg, J. J., Röckmann, T., Aerts, R., Blauw, L. G., Egger, M., Jetten, M. S., de Jong, A. E., Meisel, O. H., Rasigraf, O., and Slomp, C. P.: Methane feedbacks to the global climate system in a warmer world, Rev. Geophys., 56, 207–250, 2018.
- Delahaye, T., Landsheere, X., Pangui, E., Huet, F., Hartmann, J. M., and Tran, H.: Measurements of H₂O broadening coefficients of infrared methane lines, J. Quant. Spectrosc. Ra., 173, 40–48, 2016a.
- Delahaye, T., Maxwell, S. E., Reed, Z. D., Lin, H., Hodges, J. T., Sung, K., Devi, V. M., Warneke, T., Spietz, P., and Tran, H.: Precise methane absorption measurements in the 1.64 µm spectral region for the MERLIN mission, J. Geophys. Res.-Atmos., 121, 7360–7370, 2016b.
- Delahaye, T., Ghysels, M., Hodges, J. T., Sung, K., Armante, R., and Tran, H.: Measurement and modeling of air-broadened methane absorption in the MERLIN spectral region at low temperatures, J. Geophys. Res.-Atmos., 124, 3556–3564, 2019.
- DiGangi, J. P., Choi, Y., Nowak, J. B., Halliday, H. S., Yang, M. M., Baier, B. C., and Sweeney, C.: ACT-America: L2 In Situ Atmospheric CO2, CO, CH4, and O3 Concentrations, Eastern USA, ORNL DAAC [data set], Oak Ridge, Tennessee, USA, https://doi.org/10.3334/ORNLDAAC/1556, 2018.
- DiGangi, J. P., Choi, Y., Nowak, J. B., Halliday, H. S., Diskin, G. S., Feng, S., Barkley, Z. R., Lauvaux, T., Pal, S., Davis, K. J.

and Baier, B. C.: Seasonal variability in local carbon dioxide biomass burning sources over central and eastern US using airborne in situ enhancement ratios, J. Geophys. Res.-Atmos., 126, e2020JD034525, https://doi.org/10.1029/2020JD034525, 2021.

- Dobler, J. T., Harrison, F. W., Browell, E. V., Lin, B., McGregor, D., Kooi, S., Choi, Y., and Ismail, S.: Atmospheric CO₂ column measurements with an airborne intensity-modulated continuous wave 1.57 μm fiber laser lidar, Appl. Optics, 52, 2874–2892, 2013.
- Dufour, E. and Bréon, F. M.: Spaceborne estimate of atmospheric CO₂ column by use of the differential absorption method: error analysis, Appl. Optics, 42, 3595–3609, 2003.
- Ehhalt, D., Prather, M., Dentener, F., Derwent, R., Dlugokencky, Edward J., Holland, E., Isaksen, I., Katima, J., Kirchhoff, V., Matson, P., Midgley, P., Wang, M., Berntsen, T., Bey, I., Brasseur, G., Buja, L., Collins, W. J., Daniel, J. S., DeMore, W. B., Derek, N., Dickerson, R., Etheridge, D., Feichter, J., Fraser, P., Friedl, R., Fuglestvedt, J., Gauss, M., Grenfell, L., Grübler, A., Harris, N., Hauglustaine, D., Horowitz, L., Jackman, C., Jacob, D., Jaeglé, L., Jain, A., Kanakidou, M., Karlsdottir, S., Ko, M., Kurylo, M., Lawrence, M., Logan, J. A., Manning, M., Mauzerall, D., McConnell, J., Mickley, L. J., Montzka, S., Muller, J. F., Olivier, J., Pickering, K., Pitari, G., Roelofs, G.-J., Rogers, H., Rognerud, B., Smith, Steven J., Solomon, S., Staehelin, J., Steele, P., Stevenson, D. S., Sundet, J., Thompson, A., van Weele, M., von Kuhlmann, R., Wang, Y., Weisenstein, D. K., Wigley, T. M., Wild, O., Wuebbles, D. J., Yantosca, R., Joos, F., and McFarland, M.: Atmospheric Chemistry and Greenhouse Gases, United States, chapter 4, https://www.ipcc.ch/report/ar3/ wg1/chapter-4-atmospheric-chemistry-and-greenhouse-gases/ (last access: 13 July 2022), 2001.
- Ehret, G., Kiemle, C., Wirth, M., Amediek, A., Fix, A., and Houweling, S.: Space-borne remote sensing of CO₂, CH₄, and N₂O by integrated path differential absorption lidar: a sensitivity analysis, Appl. Phys. B, 90, 593–608, 2008.
- Ehret, G., Bousquet, P., Pierangelo, C., Alpers, M., Millet, B., Abshire, J. B., Bovensmann, H., Burrows, J. P., Chevallier, F., Ciais, P., and Crevoisier, C.: MERLIN: A French-German space lidar mission dedicated to atmospheric methane, Remote Sens., 9, 1052, https://doi.org/10.3390/rs9101052, 2017.
- Feng, S., Lauvaux, T., Keller, K., Davis, K. J., Rayner, P., Oda, T., and Gurney, K. R.: A road map for improving the treatment of uncertainties in high-resolution regional carbon flux inverse estimates, Geophys. Res. Lett., 46, 13461–13469, 2019a.
- Feng, S., Lauvaux, T., Davis, K. J., Keller, K., Zhou, Y., Williams, C., Schuh, A. E., Liu, J., and Baker, I.: Seasonal characteristics of model uncertainties from biogenic fluxes, transport, and largescale boundary inflow in atmospheric CO₂ simulations over North America, J. Geophys. Res.-Atmos., 124, 14325–14346, 2019b.
- Fernald, F. G.: Analysis of atmospheric lidar observations: some comments, Appl. Optics, 23, 652–653, 1984.
- Fitzpatrick, F., Hansell, J., Shuman, T., and Young, J.: Laser transmitter development for a methane DIAL system, in: Novel Optical Systems, Methods, and Applications XXII, SPIE, Vol. 11105, p. 111050U, https://doi.org/10.1117/12.2532032, 2019.
- Fix, A., Quatrevalet, M., Amediek, A., and Wirth, M.: Energy calibration of integrated path differential absorption lidars, Appl. Optics, 57, 7501–7514, 2018.

- Fix, A., Amediek, A., Ehret, G., Quatrevalet, M., Wirth, M., Galkowski, M., and Gerbig, C.: On the prospects of using the cloud slicing technique in integrated path differential absorption lidars, in: 16th International Workshop on Greenhouse Gas Measurements from Space, 2–5 June 2020, https://cdn.eventsforce.net/files/ef-xnn67yq56ylu/website/9/ 3_701_andreas_fix_-short_on_the_prospects_of_using_ the_cloud_slicing_technique_in_integrated_path_differential_ absorption_lidars.pdf, last access: 30 March 2022), 2020.
- Frankenberg, C., Aben, Bergamaschi, P., Dlugokencky, E. J., Van Hees, R., Houweling, S., Van Der Meer, P., Snel, R., and Tol, P.: Global column-averaged methane mixing ratios from 2003 to 2009 as derived from SCIAMACHY: Trends and variability, J. Geophys. Res., 116, D04302, https://doi.org/10.1029/2010JD014849, 2011.
- Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., Randles, C. A., Darmenov, A., Bosilovich, M. G., Reichle, R., and Wargan, K.: The modern-era retrospective analysis for research and applications, version 2 (MERRA-2), J. Climate, 30, 5419–5454, 2017.
- Gibert, F., Flamant, P. H., Bruneau, D., and Loth, C.: Twomicrometer heterodyne differential absorption lidar measurements of the atmospheric CO₂ mixing ratio in the boundary layer, Appl. Optics, 45, 4448–4458, 2006.
- Gibert, F., Flamant, P. H., Cuesta, J., and Bruneau, D: Vertical 2-μm heterodyne differential absorption lidar measurements of mean CO₂ mixing ratio in the troposphere, J. Atmos. Ocean. Tech., 25, 1477–1497, 2008.
- Gordon, I. E., Rothman, L. S., Hill, C., Kochanov, R. V., Tan, Y., Bernath, P. F., Birk, M., Boudon, V., Campargue, A., Chance, K. V., and Drouin, B. J.: The HITRAN2016 molecular spectroscopic database, J. Quant. Spectrosc. Ra., 203, 3–69, 2017.
- Grant, W. B.: Effect of differential spectral reflectance on DIAL measurements using topographic targets, Appl. Optics, 21, 2390– 2394, 1982.
- Hair, J. W., Hostetler, C. A., Cook, A. L., Harper, D. B., Ferrare, R. A., Mack, T. L., Welch, W., Izquierdo, L. R., and Hovis, F. E.: Airborne high spectral resolution lidar for profiling aerosol optical properties, Appl. Optics, 47, 6734–6752, 2008.
- Hastings, D. A. and Dunbar, P.: Development & assessment of the global land one-km base elevation digital elevation model (GLOBE), Group, 4, 218–221, 1998.
- Heimburger, A. M., Harvey, R. M., Shepson, P. B., Stirm, B. H., Gore, C., Turnbull, J., Cambaliza, M. O., Salmon, O. E., Kerlo, A. E. M., Lavoie, T. N., and Davis, K. J.: Assessing the optimized precision of the aircraft mass balance method for measurement of urban greenhouse gas emission rates through averaging, Elementa: Science of the Anthropocene, 5, 26, https://doi.org/10.1525/elementa.134, 2017.
- Hu, H., Landgraf, J., Detmers, R., Borsdorff, T., Aan de Brugh, J., Aben, I., Butz, A., and Hasekamp, O.: Toward global mapping of methane with TROPOMI: First results and intersatellite comparison to GOSAT, Geophys. Res. Lett., 45, 3682–3689, 2018.
- Ismail, S. and Browell, E. V.: Airborne and spaceborne lidar measurements of water vapor profiles: a sensitivity analysis, Appl. Optics, 28, 3603–3615, 1989.
- Jacob, D. J., Turner, A. J., Maasakkers, J. D., Sheng, J., Sun, K., Liu, X., Chance, K., Aben, I., McKeever, J., and Frankenberg, C.: Satellite observations of atmospheric methane and

their value for quantifying methane emissions, Atmos. Chem. Phys., 16, 14371–14396, https://doi.org/10.5194/acp-16-14371-2016, 2016.

- Jervis, D., McKeever, J., Durak, B. O. A., Sloan, J. J., Gains, D., Varon, D. J., Ramier, A., Strupler, M., and Tarrant, E.: The GHGSat-D imaging spectrometer, Atmos. Meas. Tech., 14, 2127–2140, https://doi.org/10.5194/amt-14-2127-2021, 2021.
- Judd, L. M., Al-Saadi, J. A., Szykman, J. J., Valin, L. C., Janz, S. J., Kowalewski, M. G., Eskes, H. J., Veefkind, J. P., Cede, A., Mueller, M., Gebetsberger, M., Swap, R., Pierce, R. B., Nowlan, C. R., Abad, G. G., Nehrir, A., and Williams, D.: Evaluating Sentinel-5P TROPOMI tropospheric NO₂ column densities with airborne and Pandora spectrometers near New York City and Long Island Sound, Atmos. Meas. Tech., 13, 6113–6140, https://doi.org/10.5194/amt-13-6113-2020, 2020.
- Karion, A., Callahan, W., Stock, M., Prinzivalli, S., Verhulst, K. R., Kim, J., Salameh, P. K., Lopez-Coto, I., and Whetstone, J.: Greenhouse gas observations from the Northeast Corridor tower network, Earth Syst. Sci. Data, 12, 699–717, https://doi.org/10.5194/essd-12-699-2020, 2020.
- Kiemle, C., Quatrevalet, M., Ehret, G., Amediek, A., Fix, A., and Wirth, M.: Sensitivity studies for a space-based methane lidar mission, Atmos. Meas. Tech., 4, 2195–2211, https://doi.org/10.5194/amt-4-2195-2011, 2011.
- Lin, B., Nehrir, A. R., Harrison, F. W., Browell, E. V., Ismail, S., Obland, M. D., Campbell, J., Dobler, J., Meadows, B., Fan, T. F., and Kooi, S.: Atmospheric CO₂ column measurements in cloudy conditions using intensity-modulated continuous-wave lidar at 1.57 micron, Opt. Express, 23, A582–A593, 2015.
- Lin, J. C., Bares, R., Fasoli, B., Garcia, M., Crosman, E., and Lyman, S.: Declining methane emissions and steady, high leakage rates observed over multiple years in a western US oil/gas production basin, Scientific Reports, 11, 1–12, 2021.
- Lopez-Coto, I., Ren, X., Salmon, O. E., Karion, A., Shepson, P. B., Dickerson, R. R., Stein, A., Prasad, K., and Whetstone, J. R.: Wintertime CO₂, CH₄, and CO emissions estimation for the Washington, DC–Baltimore metropolitan area using an inverse modeling technique, Environ. Sci. Technol., 54, 2606– 2614, 2020.
- Lu, X., Jacob, D. J., Wang, H., Maasakkers, J. D., Zhang, Y., Scarpelli, T. R., Shen, L., Qu, Z., Sulprizio, M. P., Nesser, H., Bloom, A. A., Ma, S., Worden, J. R., Fan, S., Parker, R. J., Boesch, H., Gautam, R., Gordon, D., Moran, M. D., Reuland, F., Villasana, C. A. O., and Andrews, A.: Methane emissions in the United States, Canada, and Mexico: evaluation of national methane emission inventories and 2010–2017 sectoral trends by inverse analysis of in situ (GLOBALVIEWplus CH₄ ObsPack) and satellite (GOSAT) atmospheric observations, Atmos. Chem. Phys., 22, 395–418, https://doi.org/10.5194/acp-22-395-2022, 2022.
- Lyon, D. R., Hmiel, B., Gautam, R., Omara, M., Roberts, K. A., Barkley, Z. R., Davis, K. J., Miles, N. L., Monteiro, V. C., Richardson, S. J., Conley, S., Smith, M. L., Jacob, D. J., Shen, L., Varon, D. J., Deng, A., Rudelis, X., Sharma, N., Story, K. T., Brandt, A. R., Kang, M., Kort, E. A., Marchese, A. J., and Hamburg, S. P.: Concurrent variation in oil and gas methane emissions and oil price during the COVID-19 pandemic, Atmos. Chem. Phys., 21, 6605–6626, https://doi.org/10.5194/acp-21-6605-2021, 2021.

R. A. Barton-Grimley et al.: HALO methane retrieval evaluation from ACT-America 2019

- Maasakkers, J. D., Jacob, D. J., Sulprizio, M. P., Turner, A. J., Weitz, M., Wirth, T., Hight, C., DeFigueiredo, M., Desai, M., Schmeltz, R., and Hockstad, L.: Gridded national inventory of US methane emissions, Environ. Sci. Technol., 50, 13123–13133, 2016.
- McKain, K., Down, A., Raciti, S. M., Budney, J., Hutyra, L. R., Floerchinger, C., Herndon, S. C., Nehrkorn, T., Zahniser, M. S., Jackson, R. B., and Phillips, N.: Methane emissions from natural gas infrastructure and use in the urban region of Boston, Massachusetts, P. Natl. Acad. Sci. USA, 112, 1941–1946, 2015.
- Menzies, R. T. and Tratt, D. M.: Differential laser absorption spectrometry for global profiling of tropospheric carbon dioxide: selection of optimum sounding frequencies for high-precision measurements, Appl. Optics, 42, 6569–6577, 2003.
- Myhre, G., D. Shindell, F.-M. Bréon, W. Collins, J. Fuglestvedt, J. Huang, D. Koch, J.-F. Lamarque, D. Lee, B. Mendoza, T. Nakajima, A. Robock, G. Stephens, T. Takemura and H. Zhang, 2013: Anthropogenic and Natural Radiative Forcing, in: Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, edited by: Stocker, T. F., Qin, D., Plattner, G.-K., Tignor, M., Allen, S. K., Boschung, J., Nauels, A., Xia, Y., Bex, V., and Midgley, P. M., Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, https://doi.org/10.1017/CBO9781107415324.008, 2013.
- NASEM (National Academies of Sciences, Engineering, and Medicine): Thriving on Our Changing Planet: A Decadal Strategy for Earth Observation from Space, The National Academies Press, Washington, DC, https://doi.org/10.17226/24938, 2018.
- Nehrir, A. R., Repasky, K. S., Carlsten, J. L., Obland, M. D., and Shaw, J. A.: Water vapor profiling using a widely tunable, amplified diode-laser-based differential absorption lidar (DIAL), J. Atmos. Ocean. Tech., 26, 733–745, 2009.
- Nehrir, A. R., Kiemle, C., Lebsock, M. D., Kirchengast, G., Buehler, S. A., Löhnert, U., Liu, C. L., Hargrave, P. C., Barrera-Verdejo, M., and Winker, D. M.: Emerging technologies and synergies for airborne and space-based measurements of water vapor profiles, Surv. Geophys., 38, 1445–1482, 2017.
- Nehrir, A., Notari, A., Harper, D., Fitzpatrick, F., Collins, J., Kooi, S., Antill, C., Hare, R., Barton-Grimley, R. A., Hair, J., Ferrare, R., Hostetler, C., and Welch, W.: The High Altitude Lidar Observatory (HALO): A multi-function lidar and technology test-bed for airborne and space-based measurements of water vapor and methane, NASA Earth Science Technology Office, http://www.estotechnology.us/techportfolio/pdf/additionalInfo/ 1914_Nehrir/Nehrir_ESTF2018_A1P2.pdf (last access: 13 July 2022), 2018.
- Nisbet, E. G., Dlugokencky, E. J., and Bousquet, P.: Methane on the rise again, Science, 343, 493–495, 2014.
- Nisbet, E. G., Dlugokencky, E. J., Manning, M. R., Lowry, D., Fisher, R. E., France, J. L., Michel, S. E., Miller, J. B., White, J. W. C., Vaughn, B., and Bousquet, P.: Rising atmospheric methane: 2007–2014 growth and isotopic shift, Global Biogeochem. Cy., 30, 1356–1370, 2016.
- Pal, S. and Davis, K. J.: ACT-America Campaign Catalog, ORNL DAAC, Oak Ridge, Tennessee, USA [data set], https://doi.org/10.3334/ORNLDAAC/1862, 2021.
- Plant, G., Kort, E. A., Floerchinger, C., Gvakharia, A., Vimont, I., and Sweeney, C.: Large fugitive methane emissions from urban

centers along the U.S. East Coast, Geophys. Res. Lett., 46, 8500–8507, https://doi.org/10.1029/2019GL082635, 2019.

- Ramanathan, A. K., Mao, J., Abshire, J. B., and Allan, G. R.: Remote sensing measurements of the CO₂ mixing ratio in the planetary boundary layer using cloud slicing with airborne lidar, Geophys. Res. Lett., 42, 2055–2062, 2015.
- Refaat, T. F., Ismail, S., Nehrir, A. R., Hair, J. W., Crawford, J. H., Leifer, I., and Shuman, T.: Performance evaluation of a 1.6-μm methane DIAL system from ground, aircraft and UAV platforms, Opt. Express, 21, 30415–30432, 2013.
- Refaat, T. F., Petros, M., Singh, U. N., Antill, C. W., and Remus, R. G.: High-Precision and High-Accuracy Column Dry-Air Mixing Ratio Measurement of Carbon Dioxide Using Pulsed 2-µm IPDA Lidar, IEEE T. Geosci. Remote, 58, 5804–5819, 2020.
- Richardson, S. J., Miles, N. L., Davis, K. J., Lauvaux, T., and Martins, D.: Tower measurement network of in situ CO₂, CO, and CH₄ surface in situ measurement network in support of the Indianapolis FLUX (INFLUX) Experiment, Elementa: Science of the Anthropocene, 5, 59, https://doi.org/10.1525/elementa.140, 2017.
- Riris, H., Numata, K., Li, S., Wu, S., Ramanathan, A., Dawsey, M., Mao, J., Kawa, R., and Abshire, J. B.: Airborne measurements of atmospheric methane column abundance using a pulsed integrated-path differential absorption lidar, Appl. Optics, 51, 8296–8305, 2012.
- Riris, H., Numata, K., Wu, S., Gonzalez, B., Rodriguez, M., Scott, S., Kawa, S., and Mao, J.: Methane optical density measurements with an integrated path differential absorption lidar from an airborne platform, J. Appl. Remote Sens., 11, 034001, https://doi.org/10.1117/1.JRS.11.034001, 2017.
- Scarino, A. J., Obland, M. D., Fast, J. D., Burton, S. P., Ferrare, R. A., Hostetler, C. A., Berg, L. K., Lefer, B., Haman, C., Hair, J. W., Rogers, R. R., Butler, C., Cook, A. L., and Harper, D. B.: Comparison of mixed layer heights from airborne high spectral resolution lidar, ground-based measurements, and the WRF-Chem model during CalNex and CARES, Atmos. Chem. Phys., 14, 5547–5560, https://doi.org/10.5194/acp-14-5547-2014, 2014.
- Schaefer, H., Fletcher, S. E. M., Veidt, C., Lassey, K. R., Brailsford, G. W., Bromley, T. M., Dlugokencky, E. J., Michel, S. E., Miller, J. B., Levin, I., and Lowe, D. C.: A 21st-century shift from fossilfuel to biogenic methane emissions indicated by ¹³CH₄, Science, 352, 80–84, 2016.
- Schotland, R. M.: The Determination of the Vertical Profile of Atmospheric Gases by Means of a Ground Based Optical Radar, in: Proceeding of the Third Symposium on Remote Sensing of Environment, University of Michigan, Ann Arbor, October, 1964.
- Schotland, R. M.: Errors in the lidar measurement of atmospheric gases by differential absorption, J. Appl. Meteorol., 13, 71–77, 1974.
- Schuldt, K. N., Mund, J., Luijkx, I. T., et al.: Multi-laboratory compilation of atmospheric carbon dioxide data for the period 1957–2020; obspack_co2_1_GLOBALVIEWplus_v7.0_2021-08-18, NOAA Global Monitoring Laboratory [data set], https://doi.org/10.25925/20210801, 2021.
- Sun, X., Abshire, J. B., Beck, J. D., Mitra, P., Reiff, K., and Yang, G.: HgCdTe avalanche photodiode detectors for airborne and spaceborne lidar at infrared wavelengths, Opt. Express, 25, 16589–16602, 2017.

- Tellier, Y., Pierangelo, C., Wirth, M., Gibert, F., and Marnas, F.: Averaging bias correction for the future space-borne methane IPDA lidar mission MERLIN, Atmos. Meas. Tech., 11, 5865– 5884, https://doi.org/10.5194/amt-11-5865-2018, 2018.
- Varon, D. J., Jacob, D. J., Jervis, D., and McKeever, J.: Quantifying time-averaged methane emissions from individual coal mine vents with GHGSat-D satellite observations, Environ. Sci. Technol., 54, 10246–10253, 2020.
- Vasilchenko, S., Tran, H., Mondelain, D., Kassi, S., and Campargue, A.: Accurate absorption spectroscopy of water vapor near 1.64 µm in support of the MEthane Remote LIdar missioN (MERLIN), J. Quant. Spectrosc. Ra., 235, 332–342, 2019.
- Veefkind, J. P., Aben, I., McMullan, K., Förster, H., De Vries, J., Otter, G., Claas, J., Eskes, H. J., De Haan, J. F., Kleipool, Q., and Van Weele, M.: TROPOMI on the ESA Sentinel-5 Precursor: A GMES mission for global observations of the atmospheric composition for climate, air quality and ozone layer applications, Remote Sens. Environ., 120, 70–83, 2012.
- Verhulst, K. R., Karion, A., Kim, J., Salameh, P. K., Keeling, R. F., Newman, S., Miller, J., Sloop, C., Pongetti, T., Rao, P., Wong, C., Hopkins, F. M., Yadav, V., Weiss, R. F., Duren, R. M., and Miller, C. E.: Carbon dioxide and methane measurements from the Los Angeles Megacity Carbon Project – Part 1: calibration, urban enhancements, and uncertainty estimates, Atmos. Chem. Phys., 17, 8313–8341, https://doi.org/10.5194/acp-17-8313-2017, 2017.
- Wecht, K. J., Jacob, D. J., Frankenberg, C., Jiang, Z., and Blake, D. R.: Mapping of North American methane emissions with high spatial resolution by inversion of SCIAMACHY satellite data, J. Geophys. Res.-Atmos., 119, 7741–7756, 2014a.
- Wecht, K. J., Jacob, D. J., Sulprizio, M. P., Santoni, G. W., Wofsy, S. C., Parker, R., Bösch, H., and Worden, J.: Spatially resolving methane emissions in California: constraints from the Cal-Nex aircraft campaign and from present (GOSAT, TES) and future (TROPOMI, geostationary) satellite observations, Atmos. Chem. Phys., 14, 8173–8184, https://doi.org/10.5194/acp-14-8173-2014, 2014b.

- Wei, Y., Shrestha, R., Pal, S., Gerken, T., Feng, S., Mc-Nelis, J., Singh, D., Thornton, M. M., Boyer, A. G., Shook, M. A., and Chen, G.: Atmospheric Carbon and Transport – America (ACT-America) data sets: Description, management, and delivery, Earth Space Sci., 8, e2020EA001634, https://doi.org/10.1029/2020EA001634, 2021.
- Yang, M. M., Barrick, J. D., Sweeney, C., DiGangi, J. P., and Bennett, J. R.: ACT-America: L1 Meteorological and Aircraft Navigational Data, ORNL DAAC [data set], Oak Ridge, Tennessee, USA, https://doi.org/10.3334/ORNLDAAC/1574, 2018.
- Yokota, T., Yoshida, Y., Eguchi, N., Ota, Y., Tanaka, T., Watanabe, H., and Maksyutov, S.: Global concentrations of CO₂ and CH₄ retrieved from GOSAT: First preliminary results, Sola, 5, 160–163, 2009.
- Yu, X., Millet, D. B., Wells, K. C., Henze, D. K., Cao, H., Griffis, T. J., Kort, E. A., Plant, G., Deventer, M. J., Kolka, R. K., Roman, D. T., Davis, K. J., Desai, A. R., Baier, B. C., McKain, K., Czarnetzki, A. C., and Bloom, A. A.: Aircraft-based inversions quantify the importance of wetlands and livestock for Upper Midwest methane emissions, Atmos. Chem. Phys., 21, 951– 971, https://doi.org/10.5194/acp-21-951-2021, 2021.
- Zhang, Y., Gautam, R., Pandey, S., Omara, M., Maasakkers, J. D., Sadavarte, P., Lyon, D., Nesser, H., Sulprizio, M. P., Varon, D. J., and Zhang, R.: Quantifying methane emissions from the largest oil-producing basin in the United States from space, Sci. Adv., 6, eaaz5120, https://doi.org/10.1126/sciadv.aaz5120, 2020.
- Zhang, Y., Jacob, D. J., Lu, X., Maasakkers, J. D., Scarpelli, T. R., Sheng, J.-X., Shen, L., Qu, Z., Sulprizio, M. P., Chang, J., Bloom, A. A., Ma, S., Worden, J., Parker, R. J., and Boesch, H.: Attribution of the accelerating increase in atmospheric methane during 2010–2018 by inverse analysis of GOSAT observations, Atmos. Chem. Phys., 21, 3643–3666, https://doi.org/10.5194/acp-21-3643-2021, 2021.